Retrospective Cost Model Reference Adaptive Control for Nonminimum-Phase Systems

Jesse B. Hoagg*

University of Kentucky, Lexington, Kentucky 40506-0503

Dennis S. Bernstein[†] University of Michigan, Ann Arbor, Michigan 48109-2140

DOI: 10.2514/1.57001

This paper presents a direct model reference adaptive controller for single-input/single-output discrete-time (and thus sampled-data) systems that are possibly nonminimum phase. The adaptive control algorithm requires knowledge of the nonminimum-phase zeros of the transfer function from the control to the output. This controller uses a retrospective performance, which is a surrogate measure of the actual performance, and a cumulative retrospective cost function, which is minimized by a recursive-least-squares adaptation algorithm. This paper develops the retrospective cost model reference adaptive controller and analyzes its stability.

I. Introduction

T HE objective of model reference adaptive control (MRAC) is to control an uncertain system so that it behaves like a given reference model in response to specified reference model commands. MRAC has been studied extensively for both continuous-time [1-8] and discrete-time systems [7-12]. In addition, MRAC has been extended to various classes of nonlinear systems [13]. However, the direct adaptive control results of [1-13], as well as related adaptive control techniques [14,15], are restricted to minimum-phase systems.

For nonminimum-phase systems, [16] shows that periodic control can be used, but this approach entails periods of open-loop operation. In [17], an adaptive controller is presented for systems with known nonminimum-phase zeros; however, this controller has only local convergence and stability properties. Another approach to addressing systems with nonminimum-phase zeros is to remove the nonminimum-phase zeros by relocating sensors and actuators or by using linear combinations of sensor measurements. However, constraints on the number and placement of sensors and actuators can make this approach infeasible. For example, a tail-controlled missile with its inertial measurement unit located behind the center of gravity is known to be nonminimum phase [18], and an aircraft's elevatorto-vertical-acceleration transfer function is often nonminimum phase [19].

Retrospective cost adaptive control (RCAC) is a discrete-time adaptive control technique for discrete-time (and thus sampled-data) systems that are possibly nonminimum phase [20–23]. RCAC uses a retrospective performance, which is the actual performance modified based on the difference between the actual past control inputs and the recomputed past control inputs. The structure of the retrospective performance is reminiscent of the augmented error signal presented in [1] and used in [2,4,10,13]; however, the construction and purpose of the retrospective performance differs from the augmented error signal. More specifically, the retrospective performance is constructed using knowledge of the system's nonminimum-phase

zeros, thus accounting for their presence. In contrast, the augmented error signals in [1,2,4,10,13] are used to accommodate reference models that are not strictly positive real but do not accommodate nonminimum-phase zeros in the plant.

RCAC has been demonstrated on multi-input/multi-output nonminimum-phase systems [20,21]. Furthermore, the stability of RCAC for single-input/single-output systems is analyzed in [23] for the model reference adaptive control problem and in [22] for command following and disturbance rejection. A related controller construction is used in [24] for continuous-time minimum-phase systems that have real nonminimum-phase zeros due to sampling.

The adaptive laws in [20–23] are derived by minimizing a retrospective cost, which is a quadratic function of the retrospective performance. In particular, [20] uses an instantaneous retrospective cost, which is a function of the retrospective performance at the current time and is minimized by a gradient-type adaptation algorithm. In contrast, [21] uses a recursive-least-squares (RLS) adaptation algorithm to minimize a cumulative retrospective cost that is a function of the retrospective performance at the current time step, as well as all previous time steps.

The present paper develops a retrospective cost model reference adaptive control (RC-MRAC) algorithm for discrete-time systems that are subject to unknown disturbances and potentially nonminimum phase. The reference model is assumed to satisfy a model-matching condition, where the numerator polynomial of the reference model duplicates the nonminimum-phase zeros of the open-loop system. This condition reflects the fact that the nonminimum-phase zeros of the plant cannot be moved through feedback or pole-zero cancellation. Numerical examples show that the plant's nonminimum-phase zeros need not be known exactly.

The present paper goes beyond prior work on retrospective cost adaptive control [20,21] by analyzing the stability of the closed-loop system for plants that are nonminimum phase. In addition, the present paper extends the control architecture of [20,21] to a more general MRAC architecture with unmatched disturbances. The current paper focuses on the single-input/single-output problem for clarity in the presentation of the assumptions, as well as the main stability results. Also, unlike [20], the current paper considers an RLS adaptation algorithm, as in [21]. Preliminary versions of some results in this paper are given in [22,23].

Section II of this paper describes the adaptive control problem, while Sec. III presents the RC-MRAC algorithm. Section IV presents a nonminimal-state-space realization for use in subsequent sections. Section V proves the existence of an ideal fixed-gain controller, and a closed-loop error system is constructed in Sec. VI. Section VII presents the closed-loop stability analysis. Section VIII provides



and

Received 16 November 2011; revision received 17 March 2012; accepted for publication 19 March 2012. Copyright © 2012 by the American Institute of Aeronautics and Astronautics, Inc. All rights reserved. Copies of this paper may be made for personal or internal use, on condition that the copier pay the \$10.00 per-copy fee to the Copyright Clearance Center, Inc., 222 Rosewood Drive, Danvers, MA 01923; include the code 0731-5090/12 and \$10.00 in correspondence with the CCC.

^{*}Assistant Professor, Department of Mechanical Engineering, 271 Ralph G. Anderson Building; jhoagg@engr.uky.edu. Member AIAA.

[†]Professor, Department of Aerospace Engineering, 1320 Beal Avenue; dsbaero@umich.edu. Member AIAA.

numerical examples, including a multiple-degree-of-freedom massspring-dashpot system, as well as the NASA Generic Transport Model (GTM). Finally, conclusions are given in Sec. IX.

II. Problem Formulation

Consider the discrete-time system:

$$y(k) = \sum_{i=1}^{n} -\alpha_{i} y(k-i) + \sum_{i=d}^{n} \beta_{i} u(k-i) + \sum_{i=0}^{n} \gamma_{i} w(k-i) \quad (1)$$

where $k \ge 0, \alpha_1, \dots, \alpha_n \in \mathbb{R}, \beta_d, \dots, \beta_n \in \mathbb{R}, \gamma_0, \dots, \gamma_n \in \mathbb{R}^{1 \times l_w}, y(k) \in \mathbb{R}$ is the output, $u(k) \in \mathbb{R}$ is the control, $w(k) \in \mathbb{R}^{l_w}$ is the exogenous disturbance, and the relative degree is d > 0. Furthermore, for all i < 0, u(i) = 0 and w(i) = 0, and the initial condition is $x_0 \triangleq [y(-1) \cdots y(-n)]^T \in \mathbb{R}^n$. Let **q** and **q**⁻¹ denote the forward-shift and backward-shift

operators, respectively. For all $k \ge 0$, Eq. (1) can be expressed as

$$\alpha(\mathbf{q})y(k-n) = \beta(\mathbf{q})u(k-n) + \gamma(\mathbf{q})w(k-n)$$
(2)

where $\alpha(\mathbf{q})$ and $\beta(\mathbf{q})$ are coprime, and

$$\begin{aligned} &\alpha(\mathbf{q}) \triangleq \mathbf{q}^n + \alpha_1 \mathbf{q}^{n-1} + \alpha_2 \mathbf{q}^{n-2} + \dots + \alpha_{n-1} \mathbf{q} + \alpha_n, \\ &\beta(\mathbf{q}) \triangleq \beta_d \mathbf{q}^{n-d} + \beta_{d+1} \mathbf{q}^{n-d-1} + \beta_{d+2} \mathbf{q}^{n-d-2} + \dots + \beta_{n-1} \mathbf{q} + \beta_n, \\ &\gamma(\mathbf{q}) = \mathbf{q}^n \gamma_0 + \mathbf{q}^{n-1} \gamma_1 + \mathbf{q}^{n-2} \gamma_2 + \dots + \mathbf{q} \gamma_{n-1} + \gamma_n \end{aligned}$$

Next, consider the reference model

$$\alpha_m(\mathbf{q})y_m(k-n_m) = \beta_m(\mathbf{q})r(k-n_m)$$
(3)

where $k \ge 0$, $y_m(k) \in \mathbb{R}$ is the reference model output, $r(k) \in \mathbb{R}$ is the bounded reference model command, $\alpha_m(\mathbf{q})$ is a monic asymptotically stable polynomial with degree $n_m > 0$, $\beta_m(\mathbf{q})$ is a polynomial with degree $n_m - d_m \ge 0$, where $d_m > 0$ is the relative degree of Eq. (3), and $\alpha_m(\mathbf{q})$ and $\beta_m(\mathbf{q})$ are coprime. Furthermore, for all i < 0, r(i) = 0, and the initial condition of Eq. (3) is $x_{m,0} \triangleq [y_m(-1) \quad \cdots \quad y_m(-n_m)]^T \in \mathbb{R}^{n_m}.$ Next, define the performance:

$$z(k) \stackrel{\Delta}{=} y(k) - y_m(k)$$

The goal is to develop an adaptive output-feedback controller that generates a control signal u(k) such that y(k) asymptotically follows $y_m(k)$ for all bounded reference model commands r(k) in the presence of the disturbance w(k). The goal is thus to drive the performance z(k) to zero. The following assumptions are made regarding the open-loop system (1):

Assumption 1. d is known.

Assumption 2. β_d is known.

Assumption 3. If $\zeta \in \mathbb{C}$, $|\zeta| \ge 1$, and $\beta(\zeta) = 0$, then ζ and its multiplicity are known.

Assumption 4. There exists a known integer \bar{n} such that $n \leq \bar{n}$.

The parameters $\alpha(\mathbf{q})$, $\beta(\mathbf{q})$, $\gamma(\mathbf{q})$, *n*, and x_0 are otherwise unknown.

Assumption 2 states that the first nonzero Markov parameter β_d from the control to the output is known. In discrete-time adaptive control for minimum-phase systems, it is common to assume that the sign of β_d is known and an upper bound on the magnitude of β_d is known [9,10,14,15], which are weaker assumptions than Assumption 2.

Assumption 3 implies that the nonminimum-phase zeros from the control to the output (i.e., the roots of $\beta(\mathbf{q})$ that lie on or outside the unit circle) are known. Assumption 3 is weaker than the classical direct adaptive control assumption that there are no nonminiumphase zeros from the control to the output [1-15].

While the analysis presented herein relies on Assumptions 2 and 3, numerical examples demonstrate that RC-MRAC is robust to errors in the model information assumed by Assumptions 2 and 3. More specifically, the numerical examples presented [20,25] suggest that Assumption 2 may be able to be weakened to the assumption that the sign of β_d is known and an upper bound on the magnitude of β_d is known. Furthermore, the current paper presents numerical examples that show that RC-MRAC is robust to errors in the nonminimumphase zero estimates. Additional numerical examples are presented in [25].

Next, the following assumptions are made regarding the exogenous disturbance w(k):

Assumption 5. The signal w(k) is bounded, and, for all $k \ge 0$, w(k)satisfies

$$\alpha_w(\mathbf{q})w(k) = 0$$

where $\alpha_w(\mathbf{q})$ is a nonzero monic polynomial whose roots are on the unit circle and do not coincide with the roots of $\beta(\mathbf{q})$.

Assumption 6. There exists a known integer \bar{n}_w such that $n_w \triangleq \deg \alpha_w(\mathbf{q}) \leq \bar{n}_w.$

The parameters $\alpha_w(\mathbf{q})$ and n_w are otherwise unknown, and w(k) is not assumed to be measured.

Finally, the following assumptions are made regarding the reference model (3):

Assumption 7. If $\zeta \in \mathbb{C}$, $|\zeta| \ge 1$, and $\beta(\zeta) = 0$, then $\beta_m(\zeta) = 0$ and the multiplicity of ζ with respect to $\beta(\mathbf{q})$ equals the multiplicity of ζ with respect to $\beta_m(\mathbf{q})$.

Assumption 8. $d_m \ge d$.

Assumption 9. $\alpha_m(\mathbf{q})$, $\beta_m(\mathbf{q})$, $y_m(k)$, and r(k) are known.

Assumption 7 implies that the numerator polynomial $\beta_m(\mathbf{q})$ of the reference model duplicates the plant's nonminimum-phase zeros from the control to the output. This assumption arises from the model reference architecture and reflects the fact that the nonminimumphase zeros of the plant cannot be moved through either feedback or pole-zero cancellation. Since the reference model duplicates the plant's nonminimum-phase zeros, its step response may exhibit initial undershoot or directions reversals depending on the number of positive nonminimum-phase zeros in the reference model. However, the reference model can contain additional zeros, which can be chosen to prevent initial undershoot. Furthermore, if $r(k) \equiv 0$, then the reference model (3) simplifies to $\alpha_m(\mathbf{q})y_m(k) = 0$, which does not explicitly depend on $\beta_m(\mathbf{q})$. In this case, the RC-MRAC controller does not depend on $\beta_m(\mathbf{q})$, and letting $\beta_m(\mathbf{q}) = \beta(\mathbf{q})$ trivially satisfies Assumption 7.

Now, consider the factorization of $\beta(\mathbf{q})$ given by

$$\beta(\mathbf{q}) = \beta_d \beta_u(\mathbf{q}) \beta_s(\mathbf{q}) \tag{4}$$

where $\beta_u(\mathbf{q})$ and $\beta_s(\mathbf{q})$ are monic polynomials; if $\zeta \in \mathbb{C}$, $|\zeta| \ge 1$, and $\beta(\zeta) = 0$, then $\beta_u(\zeta) = 0$ and $\beta_s(\zeta) \neq 0$; $n_u \leq n - d$ is the degree of $\beta_u(\mathbf{q})$; and $n_s \triangleq n - n_u - d$ is the degree of $\beta_s(\mathbf{q})$. Thus, Assumption 3 is equivalent to the assumption that $\beta_u(\mathbf{q})$ is known (and thus n_u is also known). Furthermore, Assumption 7 is equivalent to the assumption that $\beta_u(\mathbf{q})$ is a factor of $\beta_m(\mathbf{q})$. Thus, $\beta_m(\mathbf{q})$ has the factorization $\beta_m(\mathbf{q}) = \beta_u(\mathbf{q})\beta_r(\mathbf{q})$, where $\beta_r(\mathbf{q})$ is a known polynomial with degree $n_m - d_m - n_u$.

Retrospective Performance and the Retrospective III. **Cost Model Reference Adaptive Controller**

This section defines the retrospective performance and presents the retrospective cost model reference adaptive control (RC-MRAC) algorithm. First, define

$$r_f(k) \stackrel{\Delta}{=} \mathbf{q}^{-(n_m - d - n_u)} \beta_r(\mathbf{q}) r(k)$$

which can be computed from the known reference model command r(k) and the known polynomial $\beta_r(\mathbf{q})$. Let $n_c \ge n$, and, for all $k \ge n_c$, consider the controller

$$u(k) = \sum_{i=1}^{n_c} L_i(k) y(k-i) + \sum_{i=1}^{n_c} M_i(k) u(k-i) + N_0(k) r_f(k)$$
(5)

where, for all $i = 1, ..., n_c, L_i : \mathbb{N} \to \mathbb{R}$ and $M_i : \mathbb{N} \to \mathbb{R}$, and $N_0 :$ $\mathbb{N} \to \mathbb{R}$ are given by the adaptive laws (13) and (14) presented below. The adaptive controller presented in this section may be implemented

with positive controller order $n_c < n$, but the analysis presented in Secs. IV, V, VI, and VII requires that $n_c \ge n$. For example, we require $n_c \ge n$ to prove the existence of an ideal fixed-gain controller that drives the performance to zero. For all $k \ge n_c$, the controller (5) can be expressed as

$$u(k) = \phi^T(k)\theta(k) \tag{6}$$

where

$$\theta(k) \triangleq \begin{bmatrix} L_1(k) & \cdots & L_{n_c}(k) & M_1(k) & \cdots & M_{n_c}(k) & N_0(k) \end{bmatrix}^T$$

and, for all $k \ge n_c$,

$$\phi(k) \triangleq [y(k-1) \cdots y(k-n_c) \quad u(k-1) \cdots u(k-n_c) \quad r_f(k)]^T$$
(7)

The controller (5) cannot be implemented for nonnegative $k < n_c$ because, for nonnegative $k < n_c$, u(k) depends on the initial condition x_0 of Eq. (1), which is not assumed to be known. Therefore, for all nonnegative integers $k < n_c$, let u(k) be given by Eq. (6), where, for all nonnegative integers $k < n_c$, $\phi(k) \in \mathbb{R}^{2n_c+1}$ is chosen arbitrarily. The choice of $\phi(k)$ for $k < n_c$ impacts the transient performance of the closed-loop adaptive system. Numerical simulations suggest that letting $\phi(0) = 0$ and inserting new data at each time step as it becomes available tends to mitigate poor transient behavior.

Next, define $\bar{\alpha}_m(\mathbf{q}^{-1}) \triangleq \mathbf{q}^{-n_m} \alpha_m(\mathbf{q}), \ \bar{\beta}_m(\mathbf{q}^{-1}) \triangleq \mathbf{q}^{-n_m} \beta_m(\mathbf{q})$, and $\bar{\beta}_u(\mathbf{q}^{-1}) \triangleq \mathbf{q}^{-n_u-d} \beta_u(\mathbf{q})$. In addition, for all $k \ge 0$, define the *filtered* performance

$$z_f(k) \stackrel{\Delta}{=} \bar{\alpha}_m(\mathbf{q}^{-1})z(k) \tag{8}$$

which can be interpreted as the output of a finite-impulse-response filter whose input is z(k) and whose zeros replicate the reference model poles. For nonnegative $k < n_m$, $z_f(k)$ depends on $z(-1), \ldots, z(-n_m)$ [i.e., the difference between the initial conditions x_0 of Eq. (1) and the initial conditions $x_{m,0}$ of Eq. (3)], which are not assumed to be known. Therefore, for nonnegative $k < n_m$, $z_f(k)$ is given by (8), where the values used for $z(-1), \ldots, z(-n_m)$ are chosen arbitrarily. Furthermore, $z_f(k)$ is computable from the measurements y(k) and $y_m(k)$, as well as the known asymptotically stable polynomial $\alpha_m(\mathbf{q})$.

Now, let $\hat{\theta} \in \mathbb{R}^{2n_c+1}$ be an optimization variable used to develop the adaptive controller update equations, and, for all $k \ge 0$, define the *retrospective performance*

$$\hat{z}(\hat{\theta}, k) \triangleq z_f(k) + \beta_d [\bar{\beta}_u(\mathbf{q}^{-1})\phi(k)]^T \hat{\theta} - \beta_d \bar{\beta}_u(\mathbf{q}^{-1})u(k)$$
$$= z_f(k) + \Phi^T(k)\hat{\theta} - \beta_d \bar{\beta}_u(\mathbf{q}^{-1})u(k)$$
(9)

where the *filtered regressor* is defined by

$$\Phi(k) \triangleq \beta_d \beta_u(\mathbf{q}^{-1})\phi(k) \tag{10}$$

where, for all k < 0, $\phi(k) = 0$. The retrospective performance (9) can be interpreted as a modification to the filtered performance $z_f(k)$ based on the difference between the actual past control inputs and the recomputed past control inputs assuming that the controller parameter vector $\hat{\theta}$ was used in the past. Next, for all $k \ge 0$, define the *retrospective performance measure*:

$$z_r(k) \stackrel{\Delta}{=} \hat{z}(\theta(k), k)$$

= $z_f(k) + \beta_d [\bar{\beta}_u(\mathbf{q}^{-1})\phi(k)]^T \theta(k) - \beta_d \bar{\beta}_u(\mathbf{q}^{-1})[\phi^T(k)\theta(k)]$
(11)

Note that $\beta_d[\bar{\beta}_u(\mathbf{q}^{-1})\phi(k)]^T\theta(k)$ and $\beta_d\bar{\beta}_u(\mathbf{q}^{-1})[\phi^T(k)\theta(k)]$, which appear in Eq. (11), are not generally equal because $\mathbf{q}^{-1}[a(k)b(k)]$ is not generally equal to $[\mathbf{q}^{-1}a(k)]b(k)$. However, if $\theta(k)$ is constant, then

$$\beta_d [\beta_u(\mathbf{q}^{-1})\phi(k)]^T \theta(k) = \beta_d \beta_u(\mathbf{q}^{-1}) [\phi^T(k)\theta(k)]$$

and in this case, Eq. (11) implies that $z_r(k) = z_f(k)$, that is, the retrospective performance measure equals the filtered performance. This provides an intuitive interpretation of the RC-MRAC adaptation law, which is presented in Theorem 1 below. Specifically, the goal of RC-MRAC is to minimize $z_r(k)$ and by extension $z_f(k)$, since $z_r(k)$ can be viewed as a surrogate measure of $z_f(k)$.

To develop the RC-MRAC law, define the cumulative retrospective cost function:

$$J(\hat{\theta},k) \triangleq \sum_{i=0}^{k} \lambda^{k-i} \hat{z}^2(\hat{\theta},i) + \lambda^k [\hat{\theta} - \theta(0)]^T R[\hat{\theta} - \theta(0)] \qquad (12)$$

where $\lambda \in (0, 1]$ and $R \in \mathbb{R}^{(2n_c+1)\times(2n_c+1)}$ is positive definite. The scalar λ is a forgetting factor, which allows more recent data to be weighted more heavily than past data. The next result along with the controller (5) provides the RC-MRAC algorithm.

Theorem 1. Let $P(0) = R^{-1}$ and $\theta(0) \in \mathbb{R}^{2n_c+1}$. Then, for each $k \ge 0$, the unique global minimizer of the cumulative retrospective cost function (12) is given by

$$\theta(k+1) = \theta(k) - \frac{P(k)\Phi(k)z_r(k)}{\lambda + \Phi^T(k)P(k)\Phi(k)}$$
(13)

where

$$P(k+1) = \frac{1}{\lambda} \left[P(k) - \frac{P(k)\Phi(k)\Phi^{T}(k)P(k)}{\lambda + \Phi^{T}(k)P(k)\Phi(k)} \right]$$
(14)

Proof. Let $\overline{P}(0) = R^{-1}$, and, for all $k \ge 0$, define

$$\bar{P}(k+1) \triangleq \left[\lambda^k R + \sum_{i=0}^k \lambda^{k-i} \Phi(i) \Phi^T(i)\right]^{-1}$$
$$= [\lambda \bar{P}^{-1}(k) + \Phi(k) \Phi^T(k)]^{-1}$$

Using the matrix inversion lemma ([5], Lemma 2.1) implies that

$$\bar{P}(k+1) = \frac{1}{\lambda}\bar{P}(k) - \frac{1}{\lambda^2}\bar{P}(k)\Phi(k)$$

$$\times \left[1 + \frac{1}{\lambda}\Phi^T(k)\bar{P}(k)\Phi(k)\right]^{-1}\Phi^T(k)\bar{P}(k)$$

$$= \frac{1}{\lambda}\left[\bar{P}(k) - \frac{\bar{P}(k)\Phi(k)\Phi^T(k)\bar{P}(k)}{\lambda + \Phi^T(k)\bar{P}(k)\Phi(k)}\right]$$
(15)

Now, it follows from Eqs. (14) and (15) that P(k) and $\overline{P}(k)$ satisfy the same difference equation. Since, in addition, $P(0) = \overline{P}(0)$, it follows that $P(k) = \overline{P}(k)$.

Next, it follows from Eq. (9) that

$$J(\hat{\theta}, k) = \hat{\theta}^T \Gamma_1(k)\hat{\theta} + \Gamma_2(k)\hat{\theta} + \Gamma_3(k)$$

where

$$\Gamma_{1}(k) \triangleq \bar{P}^{-1}(k+1) = P^{-1}(k+1)$$

$$\Gamma_{2}(k) \triangleq -2\lambda^{k}\theta^{T}(0)R + 2\sum_{i=0}^{k}\lambda^{k-i}[z_{f}(i) - \beta_{d}\bar{\beta}_{u}(\mathbf{q}^{-1})u(i)]\Phi^{T}(i)$$

$$\Gamma_{3}(k) \triangleq \lambda^{k}\theta^{T}(0)R\theta(0) + \sum_{i=0}^{k}\lambda^{k-i}[z_{f}(i) - \beta_{d}\bar{\beta}_{u}(\mathbf{q}^{-1})u(i)]^{2}$$

The cost function (12) has the unique global minimizer

$$\theta(k+1) \triangleq -\frac{1}{2}\Gamma_1^{-1}(k)\Gamma_2^T(k)$$

which implies that

 $\theta(k+1)$

1770

$$= P(k+1) \left[\lambda^{k} R \theta(0) - \sum_{i=0}^{k} \lambda^{k-i} [z_{f}(i) - \beta_{d} \bar{\beta}_{u}(\mathbf{q}^{-1})u(i)] \Phi(i) \right]$$

$$= P(k+1) \left[-\frac{\lambda}{2} \Gamma_{2}^{T}(k-1) - [z_{f}(k) - \beta_{d} \bar{\beta}_{u}(\mathbf{q}^{-1})u(k)] \Phi(k) \right]$$

$$= P(k+1) [\lambda P^{-1}(k)\theta(k) - [z_{f}(k) - \beta_{d} \bar{\beta}_{u}(\mathbf{q}^{-1})u(k)] \Phi(k)]$$

Adding and subtracting $\Phi(k)\Phi^T(k)\theta(k)$ to the right-hand side and using Eq. (11) yields

$$\begin{aligned} \theta(k+1) &= P(k+1)[\lambda P^{-1}(k)\theta(k) + \Phi(k)\Phi^{T}(k)\theta(k) - \Phi(k)z_{r}(k)] \\ &= P(k+1)[P^{-1}(k+1)\theta(k) - \Phi(k)z_{r}(k)] \\ &= \theta(k) - P(k+1)\Phi(k)z_{r}(k) \end{aligned}$$
(16)

Finally, it follows from Eq. (14) that

$$P(k+1)\Phi(k) = \frac{1}{\lambda} \left[\frac{\lambda + \Phi^{T}(k)P(k)\Phi(k)}{\lambda + \Phi^{T}(k)P(k)\Phi(k)} P(k)\Phi(k) - \frac{P(k)\Phi(k)\Phi^{T}(k)P(k)\Phi(k)}{\lambda + \Phi^{T}(k)P(k)\Phi(k)} \right]$$
$$= \frac{P(k)\Phi(k)}{\lambda + \Phi^{T}(k)P(k)\Phi(k)}$$
(17)

and combining Eq. (16) with Eq. (17) yields Eq. (13).

Therefore, the RC-MRAC algorithm is given by Eqs. (6), (13), and (14), where $\phi(k)$, $\Phi(k)$, and $z_r(k)$ are given by Eqs. (7), (10), and (11), respectively. The RC-MRAC architecture is shown in Fig. 1. RC-MRAC uses the RLS-based adaptive laws (13) and (14), where P(k) is the RLS covariance matrix. The initial condition $P(0) = R^{-1}$ of the covariance matrix impacts the transient performance and convergence speed of the adaptive controller, and is the primary tuning parameter for the adaptive controller. For example, increasing the singular values of P(0) tends to increase the speed of convergence; however, convergence behavior is affected by other factors, such as the initial condition $\theta(0)$ and the persistency of excitation in $\Phi(k)$.

The remainder of this paper is devoted to analyzing the stability properties of the closed-loop adaptive system and providing numerical examples.

IV. Nonminimal-State-Space Realization

A nonminimal-state-space realization of the time-series model (1) is used to analyze the stability of the closed-loop adaptive system. The state $\phi(k)$ of this nonminimal-state-space realization consists



Fig. 1 Schematic diagram of the RC-MRAC architecture given by Eqs. (6), (13), and (14).

entirely of measured information, specifically, past values of y and u, as well as the current value of r_f . To construct this realization, define

$$\mathcal{N}_{p} \triangleq \begin{bmatrix} 0 & \cdots & 0 & 0\\ 1 & & 0 & 0\\ \vdots & \ddots & \vdots & \vdots\\ 0 & \cdots & 1 & 0 \end{bmatrix} \in \mathbb{R}^{p \times p}, \qquad E_{p} \triangleq \begin{bmatrix} 1\\ 0_{(p-1) \times 1} \end{bmatrix} \in \mathbb{R}^{p}$$

where p is a positive integer. Next, for all $k \ge n_c$, consider the $(2n_c + 1)$ th-order nonminimal-state-space realization of Eq. (1) given by

$$\phi(k+1) = \mathcal{A}\phi(k) + \mathcal{B}u(k) + \mathcal{D}_1\psi_w(k) + \mathcal{D}_r r_f(k+1)$$
(18)

$$y(k) = \mathcal{C}\phi(k) + \mathcal{D}_2\psi_w(k) \tag{19}$$

where

$$\mathcal{A} \stackrel{\Delta}{=} \mathcal{A}_{\text{nil}} + E_{2n_c+1} \mathcal{C} \in \mathbb{R}^{(2n_c+1) \times (2n_c+1)}$$
(20)

$$\mathcal{A}_{\text{nil}} \stackrel{\Delta}{=} \begin{bmatrix} \mathcal{N}_{n_c} & \mathbf{0}_{n_c \times n_c} & \mathbf{0}_{n_c \times 1} \\ \mathbf{0}_{n_c \times n_c} & \mathcal{N}_{n_c} & \mathbf{0}_{n_c \times 1} \\ \mathbf{0}_{1 \times n_c} & \mathbf{0}_{1 \times n_c} & \mathbf{0} \end{bmatrix} \in \mathbb{R}^{(2n_c+1) \times (2n_c+1)}$$
(21)

$$\mathcal{B} \triangleq \begin{bmatrix} 0_{n_c \times 1} \\ E_{n_c} \\ 0 \end{bmatrix} \in \mathbb{R}^{(2n_c+1) \times 1}$$
(22)

$$\mathcal{C} \triangleq \begin{bmatrix} -\alpha_1 & \cdots & -\alpha_n & 0_{1 \times (n_c - n)} \\ 0_{1 \times (d - 1)} & \beta_d & \cdots & \beta_n & 0_{1 \times (n_c - n + 1)} \end{bmatrix} \in \mathbb{R}^{1 \times (2n_c + 1)}$$
(23)

$$\mathcal{D}_{1} \triangleq E_{2n_{c}+1} \mathcal{D}_{2} \in \mathbb{R}^{(2n_{c}+1) \times l_{w}(n+1)}$$
$$\mathcal{D}_{2} \triangleq [\gamma_{0} \cdots \gamma_{n}] \in \mathbb{R}^{1 \times l_{w}(n+1)}$$
$$D_{r} \triangleq [0_{1 \times 2n_{c}} \quad 1]^{T} \in \mathbb{R}^{(2n_{c}+1) \times 1}$$
(24)

and $\psi_w(k) \stackrel{\Delta}{=} [w^T(k) \cdots w^T(k-n)]^T \in \mathbb{R}^{l_w(n+1)}$.

The triple $(\mathcal{A}, \mathcal{B}, \mathcal{C})$ is stabilizable and detectable but is neither controllable nor observable. In particular, $(\mathcal{A}, \mathcal{B}, \mathcal{C})$ has *n* controllable and observable eigenvalues, while $(\mathcal{A}, \mathcal{B})$ has $n_c - n + 1$ uncontrollable eigenvalues at 0, and $(\mathcal{A}, \mathcal{C})$ has $2n_c - n + 1$ unobservable eigenvalues at 0.

V. Ideal Fixed-Gain Controller

This section proves the existence of an ideal fixed-gain controller for the open-loop system (1). This controller, whose structure is illustrated in Fig. 2, is used in the next section to construct an error system for analyzing the closed-loop adaptive system. An ideal fixed-gain controller consists of four parts, specifically, a feedforward controller whose input is r_f ; a precompensator that cancels the stable zeros of the open-loop system (i.e., the roots of $\beta_s(\mathbf{q})$); an internal model of the exogenous disturbance dynamics $\alpha_w(\mathbf{q})$; and a feedback controller that stabilizes the closed loop.



Fig. 2 Schematic diagram of the closed-loop system with the ideal fixed-gain controller.

For more information on internal model control in discrete time, see [26].

For all $k \ge n_c$, consider the system (1) with $u(k) = u_*(k)$, where $u_*(k)$ is the ideal control. More precisely, for all $k \ge n_c$, consider the system

$$y_*(k) = -\sum_{i=1}^n \alpha_i y_*(k-i) + \sum_{i=d}^n \beta_i u_*(k-i) + \sum_{i=0}^n \gamma_i w(k-i)$$
(25)

where, for all $k \ge n_c$, $u_*(k)$ is given by the strictly proper ideal fixedgain controller:

$$u_*(k) = \sum_{i=1}^{n_c} L_{*,i} y_*(k-i) + \sum_{i=1}^{n_c} M_{*,i} u_*(k-i) + N_* r_f(k) \quad (26)$$

where $L_{*,1}, \ldots, L_{*,n_c} \in \mathbb{R}$, $M_{*,1}, \ldots, M_{*,n_c} \in \mathbb{R}$, $N_* \in \mathbb{R}$, and the initial condition at $k = n_c$ for Eqs. (25) and (26) is

$$\phi_{*,0} \triangleq [y_*(n_c-1) \cdots y_*(0) \ u_*(n_c-1) \cdots u_*(0) \ r_f(n_c)]^T$$

For all $k \ge n_c$, the ideal control $u_*(k)$ can be written as

$$u_*(k) = \phi_*^T(k)\theta_* \tag{27}$$

where

$$\theta_* \triangleq [L_{*,1} \quad \cdots \quad L_{*,n_c} \quad M_{*,1} \quad \cdots \quad M_{*,n_c} \quad N_*]^T$$

and

$$\phi_*(k) \stackrel{\Delta}{=} \begin{bmatrix} y_*(k-1) & \cdots & y_*(k-n_c) \\ u_*(k-1) & \cdots & u_*(k-n_c) & r_f(k) \end{bmatrix}^T$$

Therefore, it follows from Eqs. (18–24) and (27) that, for all $k \ge n_c$, the ideal closed-loop system (25) and (26), has the $(2n_c + 1)$ th order nonminimal-state-space realization

$$\phi_*(k+1) = \mathcal{A}_*\phi_*(k) + \mathcal{D}_1\psi_w(k) + \mathcal{D}_r r_f(k+1)$$
(28)

$$y_*(k) = \mathcal{C}\phi_*(k) + \mathcal{D}_2\psi_w(k) \tag{29}$$

where

$$\mathcal{A}_{*} \triangleq \mathcal{A} + \mathcal{B}\theta_{*}^{T} = \mathcal{A}_{\text{nil}} + \begin{bmatrix} E_{n_{c}}\mathcal{C} \\ E_{n_{c}}\theta_{*}^{T} \\ 0 \end{bmatrix}$$
(30)

and the initial condition is $\phi_*(n_c) = \phi_{*,0}$.

The following result guarantees the existence of an ideal fixedgain controller of the form in Eq. (26) with certain properties that are needed for the subsequent stability analysis.

Theorem 2. Let n_c satisfy

$$n_c \ge \max(n+2n_w, n_m - n_u - d) \tag{31}$$

Then there exists an ideal fixed-gain controller (26) of order n_c such that the following statements hold for the ideal closed-loop system consisting of Eqs. (25) and (26), which has the $(2n_c + 1)$ th-order nonminimal-state-space realization (28–30):

1) For all initial conditions $\phi_{*,0}$ and for all $k \ge k_0 \triangleq 2n_c + n_u + d$,

$$\bar{\alpha}_m(\mathbf{q}^{-1})y_*(k) = \bar{\beta}_m(\mathbf{q}^{-1})r(k) \tag{32}$$

and thus,

$$\bar{\alpha}_m(\mathbf{q}^{-1})y_*(k) = \bar{\alpha}_m(\mathbf{q}^{-1})y_m(k) \tag{33}$$

2) \mathcal{A}_* is asymptotically stable.

3) For all initial conditions $\phi_{*,0}$, $u_*(k)$ is bounded.

4) For all $k \ge k_0$ and all sequences e(k),

$$\beta_d \bar{\beta}_u(\mathbf{q}^{-1}) e(k) = \bar{\alpha}_m(\mathbf{q}^{-1}) \left[\sum_{i=1}^{k-n_c} \mathcal{C} \mathcal{A}_*^{i-1} \mathcal{B} e(k-i) \right]$$
(34)

The proof of Theorem 2 is in Appendix A. The lower bound on the controller order, given by Eq. (31), is a sufficient condition to guarantee the existence of an ideal fixed-gain controller. If there is no disturbance (i.e., $n_w = 0$) and the reference model is selected such that its order satisfies $n_m \le n + n_u + d$, then Eq. (31) is satisfied by a controller order greater than or equal to the order *n* of the plant.

Property 4 of Theorem 2 is a time-domain property that has the z-domain interpretation

$$C(\mathbf{z}I - \mathcal{A}_*)^{-1}\mathcal{B} = \frac{\beta_d \beta_u(\mathbf{z}) \mathbf{z}^{n_m - n_u - d}}{\alpha_m(\mathbf{z})}$$
(35)

which implies that the nonminimum-phase zeros of the closed-loop transfer function (35) are exactly the nonminimum-phase zeros of the open-loop system, that is, the roots of $\beta_u(\mathbf{q})$. Furthermore, Eq. (35) is the closed-loop transfer function from a control input perturbation e (that is, the amount that the actual control signal differs from the control signal generated by the ideal controller) to the performance z. In the subsequent sections of this paper, Eq. (34) is used to relate $z_f(k)$ and $z_r(k)$ to the controller-parameter-estimation error $\theta(k) - \theta_*$.

VI. Error System

Now, an error system is constructed using the ideal fixed-gain controller (which is not implemented) and the adaptive controller presented in Sec. III. Since n and n_w are unknown, the lower bound for the controller order n_c given by Eq. (31) is unknown. Thus, for the remainder of this paper, let n_c satisfy the lower bound

$$n_c \ge \max(\bar{n} + 2\bar{n}_w, n_m - n_u - d) \tag{36}$$

where Assumptions 1, 3, 4, 6, and 9 imply that the lower bound on n_c given by Eq. (36) is known. Furthermore, since, by Assumptions 4 and 6, $n \leq \bar{n}$ and $n_w \leq \bar{n}_w$, it follows that Eq. (36) implies Eq. (31). Next, let $\theta_* \in \mathbb{R}^{2n_c+1}$ denote the ideal fixed-gain controller given

Next, let $\theta_* \in \mathbb{R}^{2n_c+1}$ denote the ideal fixed-gain controller given by Theorem 2, and, for all $k \ge n_c$, let $\phi_*(k)$ denote the state of the ideal closed-loop system (28) and (29), where the initial condition is $\phi_{*,0} = \phi_*(n_c) = \phi(n_c)$. Furthermore, define $k_0 \triangleq 2n_c + n_u + d$. For all $k \ge n_c$, the closed-loop system consisting of Eqs. (6), (18), and (19) becomes

$$\phi(k+1) = \mathcal{A}_*\phi(k) + \mathcal{B}\phi^T(k)\tilde{\theta}(k) + \mathcal{D}_1\psi_w(k) + \mathcal{D}_r r_f(k+1)$$
(37)

$$y(k) = \mathcal{C}\phi(k) + \mathcal{D}_2\psi_w(k) \tag{38}$$

where $\tilde{\theta}(k) \triangleq \theta(k) - \theta_*$ and \mathcal{A}_* is given by Eq. (30)

Now, construct an error system by combining the ideal closed-loop system (28) and (29) with the adaptive closed-loop system (37) and (38). For all $k \ge n_c$, define the error state

$$\phi(k) \stackrel{\Delta}{=} \phi(k) - \phi_*(k)$$

and subtract Eqs. (28) and (29) from Eqs. (37) and (38) to obtain, for all $k \ge n_c$,

$$\tilde{\phi}(k+1) = \mathcal{A}_* \tilde{\phi}(k) + \mathcal{B} \phi^T(k) \tilde{\theta}(k)$$
(39)

$$\tilde{v}(k) = C\tilde{\phi}(k) \tag{40}$$

where

$$\tilde{y}(k) \stackrel{\Delta}{=} y(k) - y_*(k)$$

The following result relates $z_f(k)$ to $\theta(k)$.

Lemma 1. Consider the open-loop system (1) with the feedback (6). Then, for all initial conditions x_0 , all sequences $\theta(k)$, and all $k \ge k_0$,

$$z_f(k) = \beta_d \bar{\beta}_u(\mathbf{q}^{-1})[\phi^T(k)\bar{\theta}(k)]$$
(41)

Proof. For all $k \ge n_c$, the error system (39) and (40) has the solution

$$\tilde{y}(k) = \mathcal{C}\mathcal{A}_*^{k-n_c}\tilde{\phi}(n_c) + \sum_{i=1}^{\kappa-n_c} \mathcal{C}\mathcal{A}_*^{i-1}\mathcal{B}\phi^T(k-i)\tilde{\theta}(k-i)$$

Since $\phi_*(n_c) = \phi(n_c)$ it follows that $\tilde{\phi}(n_c) = 0$, and thus, for all $k \ge n_c$,

$$\tilde{y}(k) = \sum_{i=1}^{k-n_c} \mathcal{CA}_*^{i-1} \mathcal{B}\phi^T(k-i)\tilde{\theta}(k-i)$$

which implies that, for all $k \ge n_c + n_m$,

$$\bar{\alpha}_m(\mathbf{q}^{-1})\tilde{y}(k) = \bar{\alpha}_m(\mathbf{q}^{-1}) \left[\sum_{i=1}^{k-n_c} \mathcal{C} \mathcal{A}_*^{i-1} \mathcal{B} \phi^T(k-i) \tilde{\theta}(k-i) \right]$$

Next, it follows from property 4 of Theorem 2 with $e(k) = \phi^T(k)\overline{\theta}(k)$ that, for all $k \ge k_0$,

$$\bar{\alpha}_m(\mathbf{q}^{-1})\tilde{\mathbf{y}}(k) = \beta_d \bar{\beta}_u(\mathbf{q}^{-1})[\phi^T(k)\tilde{\theta}(k)]$$

Finally, note that

$$\bar{\alpha}_m(\mathbf{q}^{-1})\tilde{y}(k) = \bar{\alpha}_m(\mathbf{q}^{-1})y(k) - \bar{\alpha}_m(\mathbf{q}^{-1})y_*(k)$$

and it follows from statement 1 of Theorem 2 that $\bar{\alpha}_m(\mathbf{q}^{-1})$ $y_*(k) = \bar{\alpha}_m(\mathbf{q}^{-1})y_m(k)$. Therefore, for all $k \ge k_0$, $z_f(k) = \bar{\alpha}_m(\mathbf{q}^{-1})\tilde{y}(k)$, thus verifying Eq. (41).

Lemma 1 relates $z_f(k)$ to $\tilde{\theta}(k)$. Although Eq. (41) is not a linear regression in $\tilde{\theta}(k)$, the following result expresses the retrospective performance measure $z_r(k)$ as a linear regression in $\tilde{\theta}(k)$.

Lemma 2. Consider the open-loop system (1) with the feedback (6). Then, for all initial conditions x_0 , all sequences $\theta(k)$, and all $k \ge k_0$,

$$z_r(k) = \Phi^T(k)\theta(k) \tag{42}$$

Proof. Adding and subtracting $\beta_d [\bar{\beta}_u(\mathbf{q}^{-1})\phi(k)]^T \theta_*$ to the right-hand side of Eq. (11) yields, for all $k \ge 0$,

$$z_r(k) = z_f(k) - \beta_d \bar{\beta}_u(\mathbf{q}^{-1})[\phi^T(k)\tilde{\theta}(k)] + \beta_d [\bar{\beta}_u(\mathbf{q}^{-1})\phi(k)]^T \tilde{\theta}(k)$$

Next, it follows from Lemma 1 that, for all $k \ge k_0$, $z_f(k) - \beta_d \bar{\beta}_u(\mathbf{q}^{-1})[\phi^T(k)\tilde{\theta}(k)] = 0$, which implies that, for all $k \ge k_0$,

$$z_r(k) = \beta_d [\bar{\beta}_u(\mathbf{q}^{-1})\phi(k)]^T \tilde{\theta}(k) = \Phi^T(k)\tilde{\theta}(k)$$

thus verifying Eq. (42).

VII. Stability Analysis

This section analyzes the stability of the RC-MRAC algorithm (6), (13), and (14), as well as the stability of the closed-loop system. The following lemma provides the stability properties of RC-MRAC. The proof is in Appendix B.

Lemma 3. Consider the open-loop system (1) satisfying Assumptions 1–9, and the cumulative retrospective cost model reference adaptive controller (6), (13), and (14), where n_c satisfies Eq. (36). Furthermore, define

$$\eta(k) \triangleq \frac{1}{1 + \Phi^T(k)P(0)\Phi(k)} \tag{43}$$

Then, for all initial conditions x_0 and $\theta(0)$, the following properties hold:

1) $\theta(k)$ is bounded. 2) $\lim_{k\to\infty} \sum_{j=0}^k \eta(j) z_r^2(j)$ exists. 3) For all positive integers *N*,

$$\lim_{k \to \infty} \sum_{j=N}^{k} \|\theta(j) - \theta(j-N)\|^2$$

exists.

4) If $\lambda = 1$, then P(k) is bounded.

Notice that property 4 of Lemma 3 applies only if the forgetting factor $\lambda = 1$. If $\lambda < 1$ and the regressor $\Phi(k)$ is not sufficiently rich, then P(k) can grow without bound ([5], pp. 473–480; [10], pp. 224–228). In practice, this effect can be mitigated by periodically resetting the covariance matrix P(k) or by adopting the techniques discussed in [5], pp. 473–480, and [10], pp. 224–228.

Next, let $\xi_1, \ldots, \xi_{n_u} \in \mathbb{C}$ denote the n_u roots of $\beta_u(\mathbf{z})$, and define

$$M(\mathbf{z},k) \triangleq \mathbf{z}^{n_c} - M_1(k)\mathbf{z}^{n_c-1} - \dots - M_{n_c-1}(k)\mathbf{z} - M_{n_c}(k)$$

which can be interpreted as the denominator polynomial of the controller (6) at each time k. Before presenting the main result of the paper, the following additional assumption is made:

Assumption 10. There exist $\epsilon > 0$ and $k_1 > 0$ such that, for all $k \ge k_1$ and for all $i = 1, ..., n_u$, $|M(\xi_i, k)| \ge \epsilon$.

Assumption 10 asymptotically bounds the instantaneous controller poles (i.e., the roots of $M(\mathbf{z}, k)$) away from the nonminimum-phase zeros of Eq. (1). Thus, Assumption 10 implies that unstable pole-zero cancellation between the plant zeros and the controller poles does not occur asymptotically in time.

The following theorem is the main result of the paper. The proof is in Appendix C.

Theorem 3. Consider the open-loop system (1) satisfying Assumptions 1–10, and the cumulative retrospective cost model reference adaptive controller (6), (13), and (14), where n_c satisfies Eq. (36). Then, for all initial conditions x_0 and $\theta(0)$, $\theta(k)$ is bounded, u(k) is bounded, and $\lim_{k\to\infty} z(k) = 0$.

Theorem 3 invokes the assumption that there exist $\epsilon > 0$ and $k_1 > 0$ such that, for all $k \ge k_1$ and for all $i = 1, ..., n_u$, $|M(\xi_i, k)| \ge \epsilon$. This assumption cannot be verified a priori. However, the assumption $|M(\xi_i, k)| \ge \epsilon$ for some arbitrarily small $\epsilon > 0$ can be verified at each time step since $M(\xi_i, k)$ can be computed from known values (i.e., the roots of $\beta_u(\mathbf{z})$ and the controller parameter $\theta(k)$). In fact, if, for some arbitrarily small $\epsilon > 0$, the condition $|M(\xi_i, k)| \ge \epsilon$ is violated at a particular time step, then the controller parameter $\theta(k)$ can be perturbed to ensure $|M(\xi_i, k)| \ge \epsilon$. For example, $\theta(k)$ can be orthogonally projected a distance ϵ away from the hyperplane in θ space defined by the equation $M(\xi_i, k) = 0$; however, determining the direction and analyzing the stability properties of this projection is an open problem. Techniques developed to prevent pole-zero cancellation for indirect adaptive control [27] may have application to this problem. Nevertheless, numerical examples suggest that asymptotic unstable pole-zero cancellation does not occur [20,21,25].

VIII. Numerical Examples

This section presents numerical examples to demonstrate RC-MRAC. In all simulations, the adaptive controller is initialized to zero (i.e., $\theta(0) = 0$) and $\lambda = 1$. For all examples, the objective is to minimize the performance $z = y - y_m$. Unless otherwise stated, the examples rely on the plant-parameter information assumed by 1–4. No additional knowledge of the plant parameters is assumed, and no known uncertainty sets are used.

Example 1. Lyapunov-stable, nonminimum-phase system without disturbance. Consider the Lyapunov-stable-but-not-asymptotically-stable, nonminimum-phase system

$$(\mathbf{q} - 0.7)^3 (\mathbf{q}^2 + 1) y(k) = 0.25 (\mathbf{q} - 1.3)$$

 $\times (\mathbf{q} - 1 - j) (\mathbf{q} - 1 + j) u(k)$

where y(0) = -1. For this example, it follows that n = 5, $n_u = 3$, d = 2, $\beta_d = 0.25$, and $\beta_u(\mathbf{q}) = (\mathbf{q} - 1.3)(\mathbf{q} - 1 - j)(\mathbf{q} - 1 + j)$. Next, consider the reference model (3), where

$$\alpha_m(\mathbf{q}) = (\mathbf{q} - 0.5)^5, \qquad \beta_m(\mathbf{q}) = \frac{\alpha_m(1)}{\beta_u(1)} \beta_u(\mathbf{q})$$

Note that the leading coefficient of $\beta_m(\mathbf{q})$ is chosen such that the reference model has unity gain at $\mathbf{z} = 1$. Finally, let r(k) be a sequence of doublets with a period of 100 samples and an amplitude of 10.

A controller order $n_c \ge 5$ is required to satisfy Eq. (36). Let $n_c = 10$. The RC-MRAC algorithm (6), (13), and (14) is implemented in feedback with $P(0) = I_{2n_c+1}$. The closed-loop system is simulated for 500 time steps, and Fig. 3 shows the time history of y, y_m , z, and u. The closed-loop adaptive system experiences transient responses for approximately half of a period of the reference model doublet. Then RC-MRAC drives the performance $z = y - y_m$ to zero, and thus y follows y_m .

Next, the controller order n_c is increased to explore the sensitivity of the closed-loop performance to the value of n_c . For $n_c = 10, 20, \ldots, 100$, the closed-loop system is simulated, where all



Fig. 3 Lyapunov-stable, nonminimum-phase plant without disturbance. The RC-MRAC algorithm (6), (13), and (14) is implemented in feedback with $n_c = 10$, $\lambda = 1$, $P(0) = I_{2n_c+1}$, and $\theta(0) = 0$. The adaptive controller drives *z* to zero.



Fig. 4 Lyapunov-stable, nonminimum-phase plant without disturbance. The RC-MRAC algorithm (6), (13), and (14) is implemented in feedback with $n_c = 40$, $\lambda = 1$, $P(0) = I_{2n-1}$, and $\theta(0) = 0$. The closed-loop performance is comparable to that shown in Fig. 3.

parameters other than n_c are the same as above. The closed-loop performance in this example is insensitive to the choice of n_c provided that $n_c \ge 5$, which is required to satisfy Eq. (36). For this example, the worst performance is obtained by letting $n_c = 40$. Figure 4 shows the time history of y, y_m , z, and u with $n_c = 40$. Over the interval of approximately k = 30 to k = 80, the closed-loop performance shown in Fig. 4 is slightly worse

than the closed-loop performance shown in Fig. 3; however, the closed-loop performances are comparable over the rest of the time history.

Example 2. Lyapunov-stable, nonminimum-phase system with disturbance. Reconsider the Lyapunov-stable, nonminimum-phase system from Example 1 with an unknown external disturbance. More specifically, consider



Fig. 5 Lyapunov-stable, nonminimum-phase plant with disturbance. The RC-MRAC algorithm (6), (13), and (14) is implemented in feedback with $n_c = 10$, $\lambda = 1$, $P(0) = I_{2n_c+1}$, and $\theta(0) = 0$. The adaptive controller drives *z* to zero. Thus, *y* follows y_m while rejecting *w*.



Fig. 6 Lyapunov-stable, nonminimum-phase plant with disturbance and 10% error in the estimates of the nonminimum-phase zeros. The RC-MRAC algorithm (6), (13), and (14) is implemented in feedback with $n_c = 10$, $\lambda = 1$, $P(0) = I_{2n_c+1}$, and $\theta(0) = 0$. The adaptive controller yields over 70% improvement in the performance *z* relative to the open-loop performance.

$$(\mathbf{q} - 0.7)^3 (\mathbf{q}^2 + 1)y(k) = 0.25(\mathbf{q} - 1.3)(\mathbf{q} - 1 - j)$$

 $\times (\mathbf{q} - 1 + j)u(k) + (\mathbf{q} - 2)(\mathbf{q} - 0.9)w(k)$

where the external disturbance is $w(k) = 0.3 \sin(0.2\pi k)$. Notice that the disturbance-to-performance transfer function is not matched with the control-to-performance transfer function. Thus, the disturbance must be rejected through the system dynamics. Furthermore, note that no information about the disturbance is available to the adaptive controller, that is, the amplitude, frequency, and phase of the disturbance are unknown.

The controller order is $n_c = 10$, which satisfies Eq. (36). All other parameters remain the same as in Example 1. The RC-MRAC algorithm (6), (13), and (14) is implemented in feedback with $P(0) = I_{2n_c+1}$. The closed-loop system is simulated for 500 time steps, and Fig. 5 shows the time history of y, y_m , z, and u. RC-MRAC drives the performance z to zero, and thus y follows y_m while rejecting the unknown exogenous disturbance w.

Example 3. Lyapunov-stable, nonminimum-phase system with disturbance and uncertain nonminimum-phase zeros. Reconsider the Lyapunov-stable, nonminimum-phase system with disturbance from Example 2, but let the estimates of the nonminimum-phase zeros used by the controller have 10% error. Specifically, let the estimate of $\beta_u(\mathbf{q})$, which is used by the reference model as well as the adaptive

law, be given by $(\mathbf{q} - 1.43)(\mathbf{q} - 1.1 - j1.1)(\mathbf{q} - 1.1 + j1.1)$. All other parameters remain the same as in Example 2. The closed-loop system is simulated for 500 time steps, and Fig. 6 shows the time history of *y*, *y_m*, *z*, and *u*. Figure 6 shows that there is some performance degradation relative to Example 2 because the closed-loop system is unable to match the reference model as required by Assumption 7. However, the performance *z* is bounded and is reduced by over 70% relative to the open-loop performance. In this example, the error in the nonminimum-phase zero estimates can be increased to approximately 18% without causing the closed-loop performance to become unbounded.

Example 4. Stabilization of a plant that is not strongly stabilizable. Consider the unstable, nonminimum-phase system

$$\mathbf{q} (\mathbf{q} - 0.1)(\mathbf{q} - 1.2)y(k) = -2(\mathbf{q} - 1.1)u(k)$$
(44)

where y(0) = 2. The reference command and disturbance are identically zero; thus, z(k) = y(k) and the control objective is output stabilization. Note that Eq. (44) is not strongly stabilizable; that is, an unstable linear controller is required to stabilize Eq. (44) [28]. For this problem, n = 3, $n_u = 1$, d = 2, $\beta_d = -2$, and $\beta_u(\mathbf{q}) = (\mathbf{q} - 1.1)$. Let $n_c = 3$, which satisfies (36). The RC-MRAC algorithm (6), (13), and (14) is implemented in feedback with $\alpha_m(\mathbf{q}) = (\mathbf{q} - 0.1)^6$ and $P(0) = I_{2n_c+1}$. Figure 7 shows the time



Fig. 7 Stabilization of a plant that is not strongly stabilizable. The RC-MRAC algorithm (6), (13), and (14) is implemented in feedback with $n_c = 3$, $\lambda = 1$, $P(0) = I_{2n_c+1}$, and $\theta(0) = 0$. The adaptive controller forces *z* asymptotically to zero, thus stabilizing the plant, which is not strongly stabilizable.

Downloaded by Univ of Kentucky on November 5, 2012 | http://arc.aiaa.org | DOI: 10.2514/1.57001

history of *z*, *u* and the three instantaneous controller poles. The closed-loop system is simulated for 100 time steps, *z* tends to zero, and the controller poles converge. For each k > 50, the instantaneous adaptive controller has an unstable positive pole at approximately 1.96. Recall that an unstable pole is required to stabilize Eq. (44); in particular, it follows from root locus arguments that a positive pole larger than 1.2 is required to stabilize Eq. (44).

Next, assume that the nonminimum-phase zero that is located at 1.1 is uncertain. The RC-MRAC adaptive controller stabilizes the output of Eq. (44) for all estimates of the nonminimum-phase zero in the interval [1.04, 1.199]. Notice that the upper bound on this interval is constrained by the location of the unstable pole at 1.2. Figures 8 and 9 show the time history of z, u and the three instantaneous controller poles for the cases where the estimate of the nonminimum-phase zero is 1.04 and 1.199, respectively.

Example 5. Sampled-data, three-mass structure. Consider the serially connected, three-mass structure shown in Fig. 10, which is given by

$$M\ddot{q} + C\dot{q} + Kq = \mu[u \ 0 \ 0]^T + \mu[0 \ w \ 0]^T$$
(45)

where $M \triangleq \operatorname{diag}(m_1, m_2, m_3), q \triangleq [q_1 \quad q_2 \quad q_2]^T$,

$$C \triangleq \begin{bmatrix} c_1 + c_2 & -c_2 & 0 \\ -c_2 & c_2 + c_3 & -c_3 \\ 0 & -c_3 & c_3 \end{bmatrix}$$
$$K \triangleq \begin{bmatrix} k_1 + k_2 & -k_2 & 0 \\ -k_2 & k_2 + k_3 & -k_3 \\ 0 & -k_3 & k_3 \end{bmatrix}$$

u is the control, *w* is the exogenous disturbance, and the input gain is $\mu = 10^2$. For this example, the masses are $m_1 = 0.1$ kg, $m_2 = 0.2$ kg and $m_3 = 0.1$ kg; the damping coefficients are $c_1 = 5$ kg/s, $c_2 = 3$ kg/s, and $c_3 = 4$ kg/s; and the spring constants are $k_1 = 11$ kg/s², $k_2 = 12$ kg/s², and $k_3 = 5$ kg/s².

The control objective is to force the position of m_3 to follow the output y_m of a reference model. The continuous-time system (45) is sampled at 20 Hz with input provided by a zero-order hold. Thus, the sample time is $T_s = 0.05$ s. Although the continuous-time system (45) from *u* to *y* is minimum-phase [29], the sampled-data system has a nonminimum-phase sampling zero located at approximately -3.4. Thus, let $\beta_u(\mathbf{q}) = \mathbf{q} + 3.4$. In addition, d = 1, and $\beta_d = 2/45$.



Fig. 8 Stabilization of a plant that is not strongly stabilizable with error in the estimate of the nonminimum-phase zero. The RC-MRAC algorithm (6), (13), and (14) is implemented in feedback with $n_c = 3$, $\lambda = 1$, $P(0) = I_{2n_c+1}$, and $\theta(0) = 0$. The plant's nonminimum-phase zero is located at 1.1, and RC-MRAC uses an estimate of the nonminimum-phase zero given by 1.04. The adaptive controller stabilizes the plant, which is not strongly stabilizable.



Fig. 9 Stabilization of a plant that is not strongly stabilizable with error in the estimate of the nonminimum-phase zero. The RC-MRAC algorithm (6), (13), and (14) is implemented in feedback with $n_c = 3$, $\lambda = 1$, $P(0) = I_{2n_c+1}$, and $\theta(0) = 0$. The plant's nonminimum-phase zero is located at 1.1, and RC-MRAC uses an estimate of the nonminimum-phase zero given by 1.199. The adaptive controller stabilizes the plant, which is not strongly stabilizable.

Next, consider the reference model (3), where $\alpha_m(\mathbf{q}) = (\mathbf{q} - 0.3)^2$, and

$$\beta_m(\mathbf{q}) = \frac{\alpha_m(1)}{\beta_u(1)} \beta_u(\mathbf{q})$$

Furthermore, for $t = kT_s \le 8$ s, let the reference model command r(k) be a sampled sequence of 1 s doublets with amplitude 0.5 m, and, for $t = kT_s > 8$ s, let the reference model command r(k) be a sampled sinusoid with frequency 2 Hz and amplitude 1 m. Finally,



Fig. 10 A serially connected three-mass structure subjected to disturbance w and control u.

the unknown disturbance is a sampled sinusoid with frequency 3.5 Hz, amplitude 0.25 m, and a constant bias of 0.1 m. More specifically, $w(k) = 0.25 \sin(7\pi T_s k) + 0.1$.

The open-loop system is given the initial conditions $q(0) = [0.1 \ 0.2 \ 0.1]^T$ m and $\dot{q}(0) = [0 \ 0 \ 0]^T$ m/s. The RC-MRAC algorithm (6), (13), and (14) is implemented in feedback with $n_c = 16$ [which satisfies Eq. (36)] and $P(0) = 10^2 I_{2n_c+1}$. Figure 11 shows the time history of y, y_m , z, and u. The closed-loop adaptive system experiences transient responses for approximately two periods of the reference model doublet. The adaptive controller subsequently drives the performance z to zero, that is, y follows y_m and rejects w. Furthermore, at 8 s, the reference model input r is changed to the 2 Hz sinusoid, and the output y continues to follow y_m with minimal transient behavior.

Example 6. NASA's GTM. This example demonstrates RC-MRAC controlling NASA's GTM [30,31] linearized about a nominal flight condition with the following parameters:

1) Flight-path angle is 0 deg and angle of attack is 3 deg.

2) Body *x*-axis, *y*-axis, and *z*-axis velocities are 161.66, 0, and 7.12 ft/s, respectively.

3) Angular velocities in roll, pitch, and yaw are 0, 0, and 0 deg/s, respectively.

4) Latitude, longitude, and altitude are 0 deg, 0 deg, and 800 ft, respectively.



Fig. 11 Sampled-data, three-mass structure. The RC-MRAC algorithm (6), (13), and (14) is implemented in feedback with $n_c = 16$, $\lambda = 1$, $P(0) = 10^2 I_{2n_c+1}$, and $\theta(0) = 0$. The adaptive controller forces *z* asymptotically to zero; thus, the position of m_3 follows y_m while rejecting *w*. Note that *y* continues to follow the command y_m after 8 s when *r* is changes to a 2 Hz sinusoid.



Fig. 12 NASA's GTM. The RC-MRAC algorithm (6), (13), and (14) is implemented in feedback with $n_c = 20$, $\lambda = 1$, and $P(0) = 10^{20}I_{2n_c+1}$. The adaptive controller force the aircraft's altitude *y* to follow the reference model y_m .

5) Roll, pitch, and yaw angles are 0.07, 3, and 90 deg, respectively.6) Elevator, aileron, and rudder angles are 2.7, 0, and 0 deg, respectively.

RC-MRAC is implemented to control GTM's behavior from elevator to altitude. Thus, u(k) is the elevator command from its nominal value, and y(k) is the altitude deviation from its nominal value. The control objective is to force the altitude y(k) to follow the output of a reference model $y_m(k)$. The elevator dynamics are assumed to be first order with a time constant $\tau = 1/10$ ([32], p. 59). More specifically, the actual elevator deflection $u_e(t)$ is the output of the elevator dynamics

$$\tau \dot{u}_e(t) + u_e(t) = u_{\rm ZOH}(t)$$

where $u_e(0) = 0$ and $u_{\text{ZOH}}(t)$ is the zero-order hold of u(k), which is generated by RC-MRAC. The linearized elevator-to-altitude transfer function for the continuous-time GTM model has a real nonminimum-phase zero. With 50 Hz sampling, the sampled-data system has a nonminimum-phase zero located at approximately 1.706, as well as a nonminimum-phase sampling zero located at approximately -3.29. Thus, let $\beta_u(\mathbf{q}) = (\mathbf{q} - 1.706)(\mathbf{q} + 3.29)$. In addition, d = 1, and $\beta_d = 10^{-5}$.

Next, consider the reference model (3), where $\alpha_m(\mathbf{q}) = (\mathbf{q} - 0.92)^8$ and

$$\beta_m(\mathbf{q}) = \frac{\alpha_m(1)}{\beta_u(1)} \beta_u(\mathbf{q})$$

The reference model is chosen such that its gain is unity at z = 1 and its step response settles in approximately 4 s without overshoot. This reference model results in a smooth output $y_m(k)$ for the reference model command r(k), which consists of a sequence of 5 ft and 10 ft step commands.

GTM is given nonzero initial conditions relative to the nominal flight condition described above. The RC-MRAC algorithm (6), (13), and (14) is implemented in feedback with $n_c = 20$ [which satisfies Eq. (36)], $\lambda = 1$, and $P(0) = 10^{20}I_{2n_c+1}$. Note that the singular values of P(0) are 10^{20} , which allows the RC-MRAC controller to adapt quickly. Figure 12 shows the time history of the altitude y, the reference model altitude y_m , the performance $z = y - y_m$, the elevator command u, and the actual elevator deflection u_e . GTM is allowed to run in open loop for 5 s in order to demonstrate the uncontrolled response; note that the altitude drifts upward due to the nonzero initial condition and the rigid-body altitude mode (e.g., an initial altitude without bound). After 5 s, the adaptive controller is turned on, and the altitude follows the reference model after a transient period of approximately 3 to 4 s.

IX. Conclusions

The retrospective cost model reference adaptive control (RC-MRAC) algorithm for single-input/single-output discrete-time (including sampled-data) systems was shown to be effective for plants that are possibly nonminimum phase and possibly subjected to disturbances with unknown spectra. The stability analysis presented in this paper relies on knowledge of the first nonzero Markov parameter and the nonminimum-phase zeros of the plant. Numerical examples demonstrated that RC-MRAC is robust to errors in the nonminimum-phase zero estimates; however, quantification of this robustness remains an open problem. Thus, the examples demonstrated that RC-MRAC can provide both command following and disturbance rejection with limited modeling information, which need not be precisely known.

Appendix A: Proof of Theorem 2

Proof. In this proof, the ideal fixed-gain controller (26), which is depicted in Fig. 2, is constructed and shown to satisfy statements 1-4 of Theorem 2.

Since
$$r_f(k) = \mathbf{q}^{-(n_m - d - n_u)} \beta_r(\mathbf{q}) r(k)$$
, multiplying Eq. (26) by \mathbf{q}^{n_c} yields, for all $k \ge 0$,

$$M_{*}(\mathbf{q})u_{*}(k) = L_{*}(\mathbf{q})y_{*}(k) + N_{*}\beta_{r}(\mathbf{q})\mathbf{q}^{n_{1}}r(k)$$
(A1)

where

$$M_*(\mathbf{q}) = \mathbf{q}^{n_c} - M_{*,1} \mathbf{q}^{n_c - 1} - \dots - M_{*,n_c - 1} \mathbf{q} - M_{*,n_c}$$
$$L_*(\mathbf{q}) = L_{*,1} \mathbf{q}^{n_c - 1} + \dots + L_{*,n_c - 1} \mathbf{q} + L_{*,n_c}$$
$$n_1 \triangleq n_c + n_u + d - n_m$$

Note that it follows from Eq. (31) that $n_1 \ge 0$. Thus, it suffices to show that there exists $L_*(\mathbf{q})$, $M_*(\mathbf{q})$, and N_* such that statements 1–4 are satisfied.

Define
$$n_f \triangleq n_c - n_s - n_w$$
, and it follows from Eq. (31) that

$$n_f = n_c - n_s - n_w \ge \max(n_u + d + n_w, n_m - n - n_w)$$

Next, let

$$M_*(\mathbf{q}) = M_f(\mathbf{q})\alpha_w(\mathbf{q})\beta_s(\mathbf{q}) \tag{A2}$$

where $M_f(\mathbf{q})$ is a monic polynomial with degree n_f . Now, it suffices to show that there exists $L_*(\mathbf{q})$, $M_f(\mathbf{q})$, and N_* , such that statements 1–4 are satisfied.

To show statement 1, consider the closed-loop system consisting of Eqs. (25) and (A1). First, it follows from Eqs. (4) and (25) that, for all $k \ge n_c$,

$$\alpha(\mathbf{q})y_*(k) = \beta_d \beta_u(\mathbf{q})\beta_s(\mathbf{q})u_*(k) + \gamma(\mathbf{q})w(k)$$
(A3)

Next, multiplying Eq. (A3) by $M_f(\mathbf{q})\alpha_w(\mathbf{q})$ and using Eq. (A2) yields

$$M_f(\mathbf{q})\alpha_w(\mathbf{q})\alpha(\mathbf{q})y_*(k) = \beta_d\beta_u(\mathbf{q})M_*(\mathbf{q})u_*(k)$$
$$+ M_f(\mathbf{q})\alpha_w(\mathbf{q})\gamma(\mathbf{q})w(k)$$

Using Eq. (A1) yields, for all $k \ge n_c$,

$$[M_f(\mathbf{q})\alpha_w(\mathbf{q})\alpha(\mathbf{q}) - \beta_d\beta_u(\mathbf{q})L_*(\mathbf{q})]y_*(k)$$

= $\beta_d N_*\beta_u(\mathbf{q})\beta_r(\mathbf{q})\mathbf{q}^{n_1}r(k) + M_f(\mathbf{q})\alpha_w(\mathbf{q})\gamma(\mathbf{q})w(k)$ (A4)

Since $\alpha_w(\mathbf{q})$ is a scalar polynomial, it follows from Assumption 5 that

$$M_f(\mathbf{q})\alpha_w(\mathbf{q})\gamma(\mathbf{q})w(k) = M_f(\mathbf{q})\gamma(\mathbf{q})\alpha_w(\mathbf{q})w(k) = 0$$

Therefore, for all $k \ge n_c$, Eq. (A4) becomes

$$[M_f(\mathbf{q})\alpha_w(\mathbf{q})\alpha(\mathbf{q}) - \beta_d\beta_u(\mathbf{q})L_*(\mathbf{q})]y_*(k)$$

= $\beta_d N_*\beta_u(\mathbf{q})\beta_r(\mathbf{q})q^{n_1}r(k)$ (A5)

Next, let $N_* = 1/\beta_d$, and since $\beta_m(\mathbf{q}) = \beta_u(\mathbf{q})\beta_r(\mathbf{q})$, it follows from Eq. (A5) that, for all $k \ge n_c$,

$$[M_f(\mathbf{q})\alpha_w(\mathbf{q})\alpha(\mathbf{q}) - \beta_d\beta_u(\mathbf{q})L_*(\mathbf{q})]y_*(k) = \beta_m(\mathbf{q})\mathbf{q}^{n_1}r(k) \quad (A6)$$

Next, show that there exist polynomials $L_*(\mathbf{q})$ and $M_f(\mathbf{q})$ such that

$$M_f(\mathbf{q})\alpha_w(\mathbf{q})\alpha(\mathbf{q}) - \beta_d\beta_u(\mathbf{q})L_*(\mathbf{q}) = \alpha_m(\mathbf{q})\mathbf{q}^{n_1}$$

First, note that

$$\deg M_f(\mathbf{q})\alpha_w(\mathbf{q})\alpha(\mathbf{q}) = n_f + n_w + n = n_c + n_u + d$$
$$= \deg \alpha_m(\mathbf{q})\mathbf{q}^{n_1}$$

Next, the degree of $M_f(\mathbf{q})$ is n_f and the degree of $L_*(\mathbf{q})$ is at most $n_c - 1$. Since, in addition, $X(\mathbf{q}) \triangleq \alpha_w(\mathbf{q})\alpha(\mathbf{q})$ and $Y(\mathbf{q}) \triangleq -\beta_d \beta_u(\mathbf{q})$ are coprime, it follows from the Diophantine equation (see [8], Theorem A.2.5) that the roots of $M_f(\mathbf{q})X(\mathbf{q}) + Y(\mathbf{q})L_*(\mathbf{q})$ can be assigned arbitrarily by choice of $L_*(\mathbf{q})$ and $M_f(\mathbf{q})$. Therefore, there exist polynomials $L_*(\mathbf{q})$ and $M_f(\mathbf{q})$ such that

$$M_f(\mathbf{q})\alpha_w(\mathbf{q})\alpha(\mathbf{q}) - \beta_d\beta_u(\mathbf{q})L_*(\mathbf{q}) = \alpha_m(\mathbf{q})\mathbf{q}^{n_1}$$

Thus, for all $k \ge n_c$, Eq. (A6) becomes

$$\alpha_m(\mathbf{q})\mathbf{q}^{n_1}y_*(k) = \beta_m(\mathbf{q})\mathbf{q}^{n_1}r(k)$$

which implies that, for all $k \ge n_c + n_1$,

$$\alpha_m(\mathbf{q})y_*(k) = \beta_m(\mathbf{q})r(k) \tag{A7}$$

Thus, for all $k \ge k_0 \stackrel{\Delta}{=} n_c + n_1 + n_m = 2n_c + n_u + d$,

$$\bar{\alpha}_m(\mathbf{q}^{-1})y_*(k) = \beta_m(\mathbf{q}^{-1})r(k)$$

thus, confirming statement 1.

To show statement 2, note that, for all $k \ge n_c$, the closed-loop system (28) and (29) is a $(2n_c + 1)$ th-order nonminimal-state-space realization of the closed-loop system (25) and (26), which has the closed-loop characteristic polynomial

$$M_*(\mathbf{q})\alpha(\mathbf{q}) - \beta(\mathbf{q})L_*(\mathbf{q})$$

= $\beta_s(\mathbf{q})[M_f(\mathbf{q})\alpha_w(\mathbf{q})\alpha(\mathbf{q}) - \beta_d\beta_u(\mathbf{q})L_*(\mathbf{q})]$
= $\beta_s(\mathbf{q})\alpha_m(\mathbf{q})\mathbf{q}^{n_1}$

Thus, the spectrum of \mathcal{A}_* consists of the $n_c + n$ roots of $\beta_s(\mathbf{q})\alpha_m(\mathbf{q})\mathbf{q}^{n_1}$ along with $n_c - n + 1$ eigenvalues located at 0, which are exactly the uncontrollable eigenvalues of the open-loop dynamics: that is, the uncontrollable eigenvalues of $(\mathcal{A}, \mathcal{B})$. Therefore, since $\alpha_m(\mathbf{q})$ and $\beta_s(\mathbf{q})$ are asymptotically stable, it follows that \mathcal{A}_* is asymptotically stable. Thus, verifying statement 2.

To show statement 3, it follows from Assumption 5 that w(k) is bounded. Since, in addition, \mathcal{A}_* is asymptotically stable, it follows from Eq. (28) that $\phi_*(k)$ is the state of an asymptotically stable linear system with the bounded inputs $\psi_w(k)$ and $r_f(k)$. Thus, $\phi_*(k)$ is bounded. Finally, since $u_*(k)$ is a component of $\phi_*(k + 1)$, it follows that $u_*(k)$ is bounded.

To show statement 4, consider the $(2n_c + 1)$ th-order nonminimalstate-space realization (28) and (29), which, for all $k \ge n_c$, has the solution

$$y_{*}(k) = C\mathcal{A}_{*}^{k-n_{c}}\phi_{*}(n_{c}) + \sum_{i=1}^{k-n_{c}} C\mathcal{A}_{*}^{i-1}\mathcal{D}_{r}r_{f}(k-i+1) + \sum_{i=1}^{k-n_{c}} C\mathcal{A}_{*}^{i-1}\mathcal{D}_{1}\psi_{w}(k-i) + \mathcal{D}_{2}\psi_{w}(k)$$

which implies that

$$\begin{aligned} \alpha_m(\mathbf{q}) y_*(k) &= \alpha_m(\mathbf{q}) [\mathcal{C}\mathcal{A}_*^{k-n_c} \phi_*(n_c)] \\ &+ \alpha_m(\mathbf{q}) \bigg[\sum_{i=1}^{k-n_c} \mathcal{C}\mathcal{A}_*^{i-1} \mathcal{D}_r r_f(k-i+1) \bigg] \\ &+ \alpha_m(\mathbf{q}) \bigg[\sum_{i=1}^{k-n_c} \mathcal{C}\mathcal{A}_*^{i-1} \mathcal{D}_1 \psi_w(k-i) + \mathcal{D}_2 \psi_w(k) \bigg] \end{aligned}$$
(A8)

Comparing Eqs. (A7) and (A8) yields, for all $k \ge n_c + n_1$,

$$\beta_{m}(\mathbf{q})r(k) = \alpha_{m}(\mathbf{q})[\mathcal{C}\mathcal{A}_{*}^{k-n_{c}}\phi_{*}(n_{c})] + \alpha_{m}(\mathbf{q})\left[\sum_{i=1}^{k-n_{c}}\mathcal{C}\mathcal{A}_{*}^{i-1}\mathcal{D}_{r}r_{f}(k-i+1)\right] + \alpha_{m}(\mathbf{q})\left[\sum_{i=1}^{k-n_{c}}\mathcal{C}\mathcal{A}_{*}^{i-1}\mathcal{D}_{1}\psi_{w}(k-i) + \mathcal{D}_{2}\psi_{w}(k)\right]$$
(A9)

Next, for all $k \ge n_c$, consider the system (1), where u(k) consists of two components: one that is generated from the ideal controller and one that is an arbitrary sequence e(k). More precisely, for all $k \ge n_c$, consider the system

$$y_e(k) = -\sum_{i=1}^n \alpha_i y_e(k-i) + \sum_{i=d}^n \beta_i u_e(k-i) + \sum_{i=0}^n \gamma_i w(k-i)$$
(A10)

where, for all $k \ge n_c$, $u_e(k)$ is given by

$$u_e(k) = \sum_{i=1}^{n_c} L_{*,i} y_e(k-i) + \sum_{i=1}^{n_c} M_{*,i} u_e(k-i) + N_* r_f(k) + e(k)$$
(A11)

where $L_{*,1}, \ldots, L_{*,n_c}, M_{*,1}, \ldots, M_{*,n_c}, N_*$ are the ideal controller parameters, and the initial condition at $k = n_c$ for Eqs. (A10) and (A11) is

$$\phi_{e,0} = [y_e(n_c - 1) \cdots y_e(0) \quad u_e(n_c - 1) \cdots u_e(0) \quad r_f(n_c)]^T$$

Furthermore, let Eqs. (A10) and (A11) have the same initial condition as the ideal closed-loop system (25) and (26), that is, let $\phi_{e,0} = \phi_{*,0}$. For all $k \ge n_c$, Eq. (A10) implies

$$\alpha(\mathbf{q})y_e(k) = \beta(\mathbf{q})u_e(k) + \gamma(\mathbf{q})w(k)$$
(A12)

and Eq. (A11) implies

$$M_{*}(\mathbf{q})u_{e}(k) = L_{*}(\mathbf{q})y_{e}(k) + N_{*}\mathbf{q}^{n_{c}}r_{f}(k) + \mathbf{q}^{n_{c}}e(k)$$

or equivalently,

$$M_f(\mathbf{q})\alpha_w(\mathbf{q})\beta_s(\mathbf{q})u_e(k) = L_*(\mathbf{q})y_e(k)$$

+ $N_*\beta_r(\mathbf{q})\mathbf{q}^{n_1}r(k) + \mathbf{q}^{n_c}e(k)$ (A13)

Next, closing the feedback loop between Eqs. (A12) and (A13) yields, for all $k \ge n_c$,

$$\begin{split} &[M_f(\mathbf{q})\alpha_w(\mathbf{q})\alpha(\mathbf{q}) - \beta_d\beta_u(\mathbf{q})L_*(\mathbf{q})]y_e(k) \\ &= \beta_d N_*\beta_u(\mathbf{q})\beta_r(\mathbf{q})\mathbf{q}^{n_1}r(k) + M_f(\mathbf{q})\alpha_w(\mathbf{q})\gamma(\mathbf{q})w(k) \\ &+ \beta_d\beta_u(\mathbf{q})\mathbf{q}^{n_c}e(k) \end{split}$$

Since Eq. (A11) is constructed with the ideal controller parameters, it follows from earlier in this proof that

$$\begin{split} M_f(\mathbf{q})\alpha_w(\mathbf{q})\gamma(\mathbf{q})w(k) &= 0\\ M_f(\mathbf{q})\alpha_w(\mathbf{q})\alpha(\mathbf{q}) - \beta_d\beta_u(\mathbf{q})L_*(\mathbf{q}) &= \alpha_m(\mathbf{q})\mathbf{q}^{n_1}\\ N_* &= 1/\beta_d \end{split}$$

Therefore, for all $k \ge n_c$,

$$\alpha_m(\mathbf{q})\mathbf{q}^{n_1}y_e(k) = \beta_m(\mathbf{q})\mathbf{q}^{n_1}r(k) + \beta_d\beta_u(\mathbf{q})\mathbf{q}^{n_c}e(k)$$

which implies that, for all $k \ge n_c + n_1$,

$$\alpha_m(\mathbf{q})y_e(k) = \beta_m(\mathbf{q})r(k) + \beta_d\beta_u(\mathbf{q})\mathbf{q}^{n_m - n_u - d}e(k)$$
(A14)

Next, for all $k \ge n_c$, consider the $(2n_c + 1)$ th-order nonminimalstate-space realization (18–24) with the feedback (A11), which has the closed-loop representation

$$\phi_e(k+1) = \mathcal{A}_*\phi_e(k) + \mathcal{B}e(k) + \mathcal{D}_1\psi_w(k) + \mathcal{D}_rr_f(k+1),$$

$$y_e(k) = \mathcal{C}\phi_e(k) + \mathcal{D}_1\psi_w(k)$$

where the $\phi_e(k)$ has the same form as $\phi(k)$ with $y_e(k)$ and $u_e(k)$ replacing y(k) and u(k), respectively. For all $k \ge n_c$, this system has the solution

$$y_e(k) = \sum_{i=1}^{k-n_c} C\mathcal{A}_*^{i-1} \mathcal{D}_r r_f(k-i+1) + \sum_{i=1}^{k-n_c} C\mathcal{A}_*^{i-1} \mathcal{D}_1 \psi_w(k-i)$$

+ $\mathcal{D}_2 \psi_w(k) + C\mathcal{A}_*^{k-n_c} \phi_e(n_c) + \sum_{i=1}^{k-n_c} C\mathcal{A}_*^{i-1} \mathcal{B}e(k-i)$

Multiplying both sides by $\alpha_m(\mathbf{q})$ yields, for all $k \ge n_c$,

$$\alpha_{m}(\mathbf{q})y_{e}(k) = \alpha_{m}(\mathbf{q}) \left[\sum_{i=1}^{k-n_{c}} C\mathcal{A}_{*}^{i-1}\mathcal{D}_{r}r_{f}(k-i+1) + \sum_{i=1}^{k-n_{c}} C\mathcal{A}_{*}^{i-1}\mathcal{D}_{1}\psi_{w}(k-i) + \mathcal{D}_{2}\psi_{w}(k) \right] + \alpha_{m}(\mathbf{q}) \left[C\mathcal{A}_{*}^{k-n_{c}}\phi_{e}(n_{c}) \right] + \alpha_{m}(\mathbf{q}) \left[\sum_{i=1}^{k-n_{c}} C\mathcal{A}_{*}^{i-1}\mathcal{B}e(k-i) \right]$$
(A15)

Since $\phi_e(n_c) = \phi_{e,0} = \phi_{*,0} = \phi_*(n_c)$, it follows from Eqs. (A9) and (A15) that, for all $k \ge n_c + n_1$,

$$\alpha_m(\mathbf{q})y_e(k) = \beta_m(\mathbf{q})r(k) + \alpha_m(\mathbf{q}) \left[\sum_{i=1}^{k-n_c} \mathcal{C}\mathcal{A}_*^{i-1}\mathcal{B}e(k-i)\right]$$
(A16)

Finally, comparing Eqs. (A14) and (A16) yields, for all $k \ge n_c + n_1$,

$$\beta_d \beta_u(\mathbf{q}) \mathbf{q}^{n_m - n_u - d} e(k) = \alpha_m(\mathbf{q}) \left[\sum_{i=1}^{k - n_c} \mathcal{C} \mathcal{A}_*^{i-1} \mathcal{B} e(k-i) \right]$$

and multiplying both sides by \mathbf{q}^{-n_m} , yields, for all $k \ge k_0$,

$$\beta_d \bar{\beta}_u(\mathbf{q}^{-1}) e(k) = \bar{\alpha}_m(\mathbf{q}^{-1}) \left[\sum_{i=1}^{k-n_c} \mathcal{CA}_*^{i-1} \mathcal{B}e(k-i) \right]$$

thus verifying statement 4.

Appendix B: Proof of Lemma 3

Proof. Subtracting θ_* from both sides of Eq. (13) yields the estimator-error update equation

$$\tilde{\theta}(k+1) = \tilde{\theta}(k) - \frac{P(k)\Phi(k)z_r(k)}{\lambda + \Phi^T(k)P(k)\Phi(k)}$$
(B1)

Next, note from Eq. (14) that

$$P(k+1)\Phi(k) = \frac{1}{\lambda} \left[P(k) - \frac{P(k)\Phi(k)\Phi^{T}(k)P(k)}{\lambda + \Phi^{T}(k)P(k)\Phi(k)} \right] \Phi(k)$$
$$= \frac{P(k)\Phi(k)}{\lambda + \Phi^{T}(k)P(k)\Phi(k)}$$
(B2)

and thus,

$$\tilde{\theta}(k+1) = \tilde{\theta}(k) - P(k+1)\Phi(k)z_r(k).$$
(B3)

Define

$$V_P(P(k), k) \stackrel{\Delta}{=} \lambda^{-k} P^{-1}(k),$$

$$\Delta V_P(k) \stackrel{\Delta}{=} V_P(P(k+1), k+1) - V_P(P(k), k)$$

and note the RLS identity [5,7,10]

$$P^{-1}(k+1) = \lambda P^{-1}(k) + \Phi(k)\Phi^{T}(k)$$
(B4)

Evaluating $\Delta V_P(k)$ along the trajectories of Eq. (B4) yields

$$\Delta V_P(k) = \frac{1}{\lambda^{k+1}} [\lambda P^{-1}(k) + \Phi(k) \Phi^T(k)] - \frac{1}{\lambda^k} P^{-1}(k)$$

= $\lambda^{-k-1} \Phi(k) \Phi^T(k)$ (B5)

Since $P^{-1}(0)$ is positive definite and ΔV_P is positive semidefinite, it follows that, for all $k \ge 0$, $V_P(P(k), k)$ is positive definite and $V_P(P(k), k) \ge V_P(P(k-1), k-1)$. Therefore, for all $k \ge 0$, $0 < V_P(P(0), 0) \le V_P(P(k), k)$, which implies that

$$0 < \lambda^k P(k) \le P(0) \tag{B6}$$

If $\lambda = 1$, then Eq. (B6) implies that P(k) is bounded, which verifies statement 4.

Next, define the positive-definite Lyapunov-like function,

$$V_{\tilde{\theta}}(\tilde{\theta}(k), P(k), k) \triangleq \tilde{\theta}^T(k) V_P(P(k), k) \tilde{\theta}(k)$$

and define the Lyapunov-like difference

$$\Delta V_{\tilde{\theta}}(k) \triangleq V_{\tilde{\theta}}(\tilde{\theta}(k+1), P(k+1), k+1) - V_{\tilde{\theta}}(\tilde{\theta}(k), P(k), k)$$
(B7)

Evaluating $\Delta V_{\bar{\theta}}(k)$ along the trajectories of the estimator-error system (B3) and using Eq. (B5) yields

$$\begin{split} \Delta V_{\tilde{\theta}}(k) &= \tilde{\theta}^{T}(k) \Delta V_{P}(k) \tilde{\theta}(k) - 2\lambda^{-k-1} z_{r}(k) \Phi^{T}(k) \tilde{\theta}(k) \\ &+ \lambda^{-k-1} z_{r}^{2}(k) \Phi^{T}(k) P(k+1) \Phi(k) \\ &= \lambda^{-k-1} [\tilde{\theta}^{T}(k) \Phi(k) \Phi^{T}(k) \tilde{\theta}(k) - 2 z_{r}(k) \Phi^{T}(k) \tilde{\theta}(k) \\ &+ z_{r}^{2}(k) \Phi^{T}(k) P(k+1) \Phi(k)] \end{split}$$

Next, it follows from Lemma 2 and Eq. (B2) that, for all $k \ge k_0$,

$$\begin{split} \Delta V_{\hat{\theta}}(k) &= -\lambda^{-k-1} z_r^2(k) (1 - \Phi^T(k) P(k+1) \Phi(k)) \\ &= -\lambda^{-k-1} z_r^2(k) \left(1 - \frac{\Phi^T(k) P(k) \Phi(k)}{\lambda + \Phi^T(k) P(k) \Phi(k)} \right) \\ &= -\lambda^{-k-1} z_r^2(k) \frac{\lambda}{\lambda + \Phi^T(k) P(k) \Phi(k)} \\ &= -\bar{\eta}(k) z_r^2(k) \end{split}$$
(B8)

where

$$\bar{\eta}(k) \triangleq \frac{1}{\lambda^{k+1} + \lambda^k \Phi^T(k) P(k) \Phi(k)}$$
(B9)

Since $V_{\tilde{\theta}}$ is a positive-definite radially unbounded function of $\hat{\theta}(k)$ and, for $k \ge k_0$, $\Delta V_{\tilde{\theta}}(k)$ is nonpositive, it follows that $\tilde{\theta}(k)$ is bounded and thus $\theta(k)$ is bounded. Thus, verifying statement 1. To show statement 2, first show that

To show statement 2, first show that

$$\lim_{k\to\infty}\sum_{j=k_0}^{\kappa}\Delta V_{\tilde{\theta}}(j)$$

exists. Since $V_{\tilde{\theta}}$ is positive definite, and, for all $k \ge k_0$, $\Delta V_{\tilde{\theta}}(k)$ is nonpositive, it follows from Eq. (B7) that

$$0 \leq -\lim_{k \to \infty} \sum_{j=k_0}^{k} \Delta V_{\bar{\theta}}(j)$$

= $V_{\bar{\theta}}(\tilde{\theta}(k_0), P(k_0), k_0) - \lim_{k \to \infty} V_{\bar{\theta}}(\tilde{\theta}(k), P(k), k)$
 $\leq V_{\bar{\theta}}(\tilde{\theta}(k_0), P(k_0), k_0)$

where the upper and lower bounds imply that both limits exist. Since

$$\lim_{k\to\infty}\sum_{j=k_0}^k \Delta V_{\tilde{\theta}}(j)$$

exists, Eq. (B8) implies that

$$\lim_{k\to\infty}\sum_{j=k_0}^k \bar{\eta}(j) z_r^2(j)$$

exists, and thus

$$\lim_{k\to\infty}\sum_{j=0}^k \bar{\eta}(j)z_r^2(j)$$

exists. Since, for all $k \ge 0$, $\lambda^{k+1} \le 1$ and $\lambda^k P(k) \le P(0)$, it follows from Eqs. (43) and (B9) that, for all $k \ge 0$, $\eta(k) \le \overline{\eta}(k)$, which implies that

$$\lim_{k \to \infty} \sum_{j=0}^k \eta(j) z_r^2(j) \le \lim_{k \to \infty} \sum_{j=0}^k \bar{\eta}(j) z_r^2(j)$$

Thus,

$$\lim_{k\to\infty}\sum_{j=0}^k\eta(j)z_r^2(j)$$

exists, which verifies statement 2.

To show statement 3, first show that

$$\lim_{k \to \infty} \sum_{j=0}^{k} \|\theta(j+1) - \theta(j)\|^2$$

exists. It follows from Eqs. (B1) and (B6) that

$$\begin{split} &\sum_{j=0}^{\infty} \|\theta(j+1) - \theta(j)\|^2 = \sum_{j=0}^{\infty} \left\| \frac{P(j)\Phi(j)z_r(j)}{\lambda + \Phi^T(j)P(j)\Phi(j)} \right\|^2 \\ &= \sum_{j=0}^{\infty} z_r^2(j) \frac{\Phi^T(j)P^2(j)\Phi(j)}{[\lambda + \Phi^T(j)P(j)\Phi(j)]^2} \\ &= \sum_{j=0}^{\infty} \bar{\eta}(j)z_r^2(j) \left(\frac{\lambda^j \Phi^T(j)P^2(j)\Phi(j)}{\lambda + \Phi^T(j)P(j)\Phi(j)} \right) \\ &\leq \sum_{j=0}^{\infty} \bar{\eta}(j)z_r^2(j) \|\lambda^j P(j)\|_F \left(\frac{\Phi^T(j)P(j)\Phi(j)}{\lambda + \Phi^T(j)P(j)\Phi(j)} \right) \\ &\leq \|P(0)\|_F \sum_{j=0}^{\infty} \bar{\eta}(j)z_r^2(j) \left(\frac{\Phi^T(j)P(j)\Phi(j)}{\lambda + \Phi^T(j)P(j)\Phi(j)} \right) \end{split}$$

where $\|\cdot\|_{F}$ denotes the Frobenius norm. Next, note that, for all $k \ge 0$,

$$\frac{\Phi^T(k)P(k)\Phi(k)}{\lambda + \Phi^T(k)P(k)\Phi(k)} \le 1$$

which implies that

$$\lim_{k \to \infty} \sum_{j=0}^{k} \|\theta(j+1) - \theta(j)\|^2 \le \|P(0)\|_F \lim_{k \to \infty} \sum_{j=0}^{k} \bar{\eta}(j) z_r^2(j)$$
(B10)

Since

$$\lim_{k\to\infty}\sum_{j=0}^k \bar{\eta}(j)z_r^2(j)$$

exists, it follows from Eq. (B10) that

$$\lim_{k \to \infty} \sum_{j=0}^k \|\theta(j+1) - \theta(j)\|^2$$

exists. Next, let N be a positive integer and note that

$$\begin{split} \sum_{j=N}^{\infty} \|\theta(j) - \theta(j-N)\|^2 &= \sum_{j=N}^{\infty} \|\theta(j) - \theta(j-1) + \theta(j-1) \\ &- \theta(j-2) + \dots + \theta(j-N+1) - \theta(j-N)\|^2 \\ &\leq \sum_{j=N}^{\infty} (\|\theta(j) - \theta(j-1)\| + \|\theta(j-1) - \theta(j-2)\| \\ &+ \dots + \|\theta(j-N+1) - \theta(j-N)\|)^2 \\ &\leq 2^{N-1} \sum_{j=N}^{\infty} (\|\theta(j) - \theta(j-1)\|^2 + \|\theta(j-1) - \theta(j-2)\|^2 \\ &+ \dots + \|\theta(j-N+1) - \theta(j-N)\|^2) \end{split}$$
(B11)

Since all of the limits on the right-hand side of Eq. (B11) exist, it follows that

$$\lim_{k \to \infty} \sum_{j=N}^{k} \|\theta(j) - \theta(j-N)\|^2$$

exists. This verifies statement 3.

Appendix C: Proof of Theorem 3

Proof. It follows from statement 1 of Lemma 3 that $\theta(k)$ is bounded. To prove the remaining properties, for all $k \ge k_0$, define the ideal filtered regressor

$$\Phi_*(k) \stackrel{\Delta}{=} \beta_d \bar{\beta}_u(\mathbf{q}^{-1}) \phi_*(k) \tag{C1}$$

and the filtered regressor error

$$\tilde{\Phi}(k) \stackrel{\Delta}{=} \Phi(k) - \Phi_*(k) = \beta_d \bar{\beta}_u(\mathbf{q}^{-1}) \tilde{\phi}(k) \tag{C2}$$

Next, apply the operator $\beta_d \bar{\beta}_u(\mathbf{q}^{-1})$ to Eq. (39) and use Lemma 1 to obtain the filtered error system

$$\tilde{\Phi}(k+1) = \mathcal{A}_* \tilde{\Phi}(k) + \mathcal{B}\beta_d \bar{\beta}_u (\mathbf{q}^{-1}) [\phi^T(k) \bar{\theta}(k)]$$

= $\mathcal{A}_* \tilde{\Phi}(k) + \mathcal{B}z_f(k)$ (C3)

which is defined for all $k \ge k_0$. Next, define the quadratic function

$$J(\tilde{\Phi}(k)) \triangleq \tilde{\Phi}^T(k) \mathcal{P}\tilde{\Phi}(k) \tag{C4}$$

where $\mathcal{P} > 0$ satisfies the discrete-time Lyapunov equation $\mathcal{P} = \mathcal{A}_*^T \mathcal{P} \mathcal{A}_* + \mathcal{Q} + \alpha I$, where $\mathcal{Q} > 0$ and $\alpha > 0$. Note that \mathcal{P} exists since A_* is asymptotically stable. Defining

$$\Delta J(k) \stackrel{\Delta}{=} J(\tilde{\Phi}(k+1)) - J(\tilde{\Phi}(k)) \tag{C5}$$

it follows from Eq. (C3) that, for all $k \ge k_0$,

$$\Delta J(k) = -\tilde{\Phi}^{T}(k)(\mathcal{Q} + \alpha I)\tilde{\Phi}(k) + \tilde{\Phi}^{T}(k)\mathcal{A}_{*}^{T}\mathcal{P}\mathcal{B}z_{f}(k)$$

$$+ z_{f}(k)\mathcal{B}^{T}\mathcal{P}\tilde{\mathcal{A}}_{*}\tilde{\Phi}(k) + z_{f}^{2}(k)\mathcal{B}^{T}\mathcal{P}\mathcal{B}$$

$$\leq -\tilde{\Phi}^{T}(k)(\mathcal{Q} + \alpha I)\tilde{\Phi}(k)$$

$$+ z_{f}^{2}(k)\mathcal{B}^{T}\mathcal{P}\mathcal{B} + \alpha\tilde{\Phi}^{T}(k)\tilde{\Phi}(k)$$

$$+ \frac{1}{\alpha}z_{f}^{2}(k)\mathcal{B}^{T}\mathcal{P}\mathcal{A}_{*}\mathcal{A}_{*}^{T}\mathcal{P}\mathcal{B}$$

$$= -\tilde{\Phi}^{T}(k)\mathcal{Q}\tilde{\Phi}(k) + \sigma_{1}z_{f}^{2} \qquad (C6)$$

where $\sigma_1 \stackrel{\Delta}{=} \mathcal{B}^T \mathcal{P} \mathcal{B} + \frac{1}{\alpha} \mathcal{B}^T \mathcal{P} \mathcal{A}_* \mathcal{A}_*^T \mathcal{P} \mathcal{B}$. Now, consider the positive-definite, radially unbounded Lyapunov-like function:

$$V(\tilde{\Phi}(k)) \triangleq \ln(1 + a_1 J(\tilde{\Phi}(k)))$$

where $a_1 > 0$ is specified below. The Lyapunov-like difference is thus given by

$$\Delta V(k) \triangleq V(\Phi(k+1)) - V(\Phi(k))$$

For all $k \ge k_0$, evaluating $\Delta V(k)$ along the trajectories of Eq. (C3) yields

$$\Delta V(k) = \ln[1 + a_1 J(\Phi(k)) + a_1 \Delta J(k)] - \ln[1 + a_1 J(\Phi(k))]$$

=
$$\ln\left(1 + \frac{a_1 \Delta J(k)}{1 + a_1 J(\tilde{\Phi}(k))}\right)$$
(C7)

Since, for all x > 0, $l_n x \le x - 1$, and using Eq. (C6) yields

$$\begin{split} \Delta V(k) &\leq a_1 \frac{\Delta J(k)}{1 + a_1 J(\tilde{\Phi}(k))} \\ &\leq -a_1 \frac{\tilde{\Phi}^T(k) \mathcal{Q} \tilde{\Phi}(k)}{1 + a_1 \tilde{\Phi}^T(k) \mathcal{P} \tilde{\Phi}(k)} + a_1 \sigma_1 \frac{z_f^2(k)}{1 + a_1 \tilde{\Phi}^T(k) \mathcal{P} \tilde{\Phi}(k)} \\ &\leq -W(\tilde{\Phi}(k)) + a_1 \sigma_1 \ell^2(k) \end{split}$$
(C8)

where

$$W(\tilde{\Phi}(k)) \triangleq a_1 \frac{\tilde{\Phi}^T(k) \mathcal{Q}\tilde{\Phi}(k)}{1 + a_1 \tilde{\Phi}^T(k) \mathcal{P}\tilde{\Phi}(k)}$$
(C9)

$$\ell(k) \triangleq \frac{z_f(k)}{\sqrt{1 + a_1 \lambda_{\min}(\mathcal{P})\tilde{\Phi}^T(k)\tilde{\Phi}(k)}}$$
(C10)

Now, show that

$$\lim_{k\to\infty}\sum_{j=0}^k \ell^2(j)$$

exists. First, write $\beta_u(\mathbf{q})$ as

$$\beta_u(\mathbf{q}) = \beta_{u,0}\mathbf{q}^{n_u} + \beta_{u,1}\mathbf{q}^{n_u-1} + \dots + \beta_{u,n_u-1}\mathbf{q} + \beta_{u,n_u}$$

where $\beta_{u,0} = 1$ and $\beta_{u,1}, \dots, \beta_{u,n_u} \in \mathbb{R}$. It follows from Eq. (11) that, for all $k \ge k_0$,

$$z_f(k) = z_r(k) - \beta_d \sum_{i=d}^{n_u+d} \beta_{u,i-d} \phi^T(k-i) [\theta(k) - \theta(k-i)] \quad (C11)$$

Using Eqs. (C10) and (C11) yields, for all $k \ge k_0$,

$$|\ell(k)| \leq \frac{|z_r(k)| + |\beta_d| (\sum_{i=d}^{n_u+d} |\beta_{u,i-d}| \|\phi(k-i)\| \|\theta(k) - \theta(k-i)\|)}{\sqrt{1 + a_1 \lambda_{\min}(\mathcal{P})\tilde{\Phi}^T(k)\tilde{\Phi}(k)}}$$

It follows from Lemma 3 that $\theta(k)$ is bounded and $\lim_{k\to\infty} \|\theta(k) - \theta(k-1)\| = 0$. Therefore, Lemma 4 in Appendix D implies that there exist $k_2 \ge k_0 > 0$, $c_1 > 0$, and $c_2 > 0$, such that, for all $k \ge k_2$ and all $i = d, \ldots, n_u + d$, $\|\phi(k-i)\| \le c_1 + c_2 \|\Phi(k)\|$. In addition, note that $\|\Phi(k)\| = \|\tilde{\Phi}(k) + \Phi_*(k)\| \le \|\tilde{\Phi}(k)\| + \|\Phi_*(k)\| \le \|\tilde{\Phi}(k)\| + \Phi_{*,\max}$, where $\Phi_{*,\max} \triangleq \sup_{k\ge 0} \|\Phi_*(k)\|$ exists because Φ_* is bounded. Therefore, for all $k \ge k_2$, $\|\phi(k-i)\| \le c_1 + c_2 \Phi_{*,\max} + c_2 \|\tilde{\Phi}(k)\|$, which implies that

Next, show that $a_1 > 0$ can be chosen such that the first term of Eq. (C12) is less than a constant times $\sqrt{\eta(k)}|z_r(k)|$, which is square summable according to statement 2 of Lemma 3. It follows from Eq. (43) that

$$\frac{1}{\eta(k)} = 1 + \Phi^{T}(k)P(0)\Phi(k)$$

$$\leq 1 + \lambda_{\max}(P(0))[\tilde{\Phi}(k) + \Phi_{*}(k)]^{T}[\tilde{\Phi}(k) + \Phi_{*}(k)]$$

$$\leq 1 + \lambda_{\max}(P(0))[2\tilde{\Phi}^{T}(k)\tilde{\Phi}(k) + 2\Phi_{*}^{T}(k)\Phi_{*}(k)]$$

$$\leq 1 + 2\lambda_{\max}(P(0))\Phi_{*,\max}^{2} + 2\lambda_{\max}(P(0))\tilde{\Phi}^{T}(k)\tilde{\Phi}(k)$$

$$= c_{6}[1 + a_{1}\lambda_{\min}(\mathcal{P})\tilde{\Phi}^{T}(k)\tilde{\Phi}(k)]$$

where $a_1 \triangleq \frac{2\lambda_{\max}(P(0))}{\lambda_{\min}(P)[1+2\lambda_{\max}(P(0))\Phi^2_{*,\max}]} > 0$ and $c_6 \triangleq 1 + 2\lambda_{\max}(P(0)) \Phi^2_{*,\max} > 0$. Therefore,

$$\frac{1}{\sqrt{1+a_1\lambda_{\min}(\mathcal{P})\tilde{\Phi}^T(k)\tilde{\Phi}(k)}} \le \sqrt{c_6}\sqrt{\eta(k)}$$

which together with Eq. (C12) implies that, for all $k \ge k_2$,

$$|\ell(k)| \le \sqrt{c_6} \sqrt{\eta(k)} |z_r(k)| + c_5 \sum_{i=d}^{n_y+d} ||\theta(k) - \theta(k-i)|$$

Therefore, for all $k \ge k_2$,

$$\ell^{2}(k) \leq \left[\sqrt{c_{6}}\sqrt{\eta(k)}|z_{r}(k)| + c_{5}\sum_{i=d}^{n_{u}+d} \left\|\theta(k) - \theta(k-i)\right\|\right]^{2}$$

$$\leq 2c_{6}\eta(k)z_{r}^{2}(k) + 2c_{5}^{2}\left[\sum_{i=d}^{n_{u}+d} \left\|\theta(k) - \theta(k-i)\right\|\right]^{2}$$

$$\leq 2c_{6}\eta(k)z_{r}^{2}(k) + 2^{n_{u}+1}c_{5}^{2}\sum_{i=d}^{n_{u}+d} \left\|\theta(k) - \theta(k-i)\right\|^{2} \quad (C13)$$

It follows from statement 2 of Lemma 3 that

$$\lim_{k\to\infty}\sum_{j=0}^k\eta(j)z_r^2(j)$$

exists. Furthermore, it follows from statement 3 of Lemma 3 that, for all $i = d, ..., n_u + d$,

$$\begin{split} |\ell(k)| &\leq \frac{|z_r(k)| + |\beta_d| (c_1 + c_2 \Phi_{*,\max} + c_2 \|\tilde{\Phi}(k)\|) (\sum_{i=d}^{n_u+d} |\beta_{u,i-d}| \|\theta(k) - \theta(k-i)\|)}{\sqrt{1 + a_1 \lambda_{\min}(\mathcal{P}) \tilde{\Phi}^T(k) \tilde{\Phi}(k)}} \\ &\leq \frac{|z_r(k)|}{\sqrt{1 + a_1 \lambda_{\min}(\mathcal{P}) \tilde{\Phi}^T(k) \tilde{\Phi}(k)}} + \frac{c_3 + c_4 \|\tilde{\Phi}(k)\|}{\sqrt{1 + a_1 \lambda_{\min}(\mathcal{P}) \tilde{\Phi}^T(k) \tilde{\Phi}(k)}} \left(\sum_{i=d}^{n_u+d} \|\theta(k) - \theta(k-i)\|\right) \end{split}$$

where $c_3 \triangleq (c_1 + c_2 \Phi_{*,\max}) |\beta_d| (\max_{0 \le i \le n_u} |\beta_{u,i}|) > 0$ and $c_4 \triangleq c_2 |\beta_d| (\max_{0 \le i \le n_u} |\beta_{u,i}|) > 0$. Next, note that $\frac{1}{\sqrt{1 + a_1 \lambda_{\min}(\mathcal{P})} \tilde{\Phi}^T(k) \tilde{\Phi}(k)} \le 1$ and $\frac{\|\tilde{\Phi}(k)\|}{\sqrt{1 + a_1 \lambda_{\min}(\mathcal{P})} \tilde{\Phi}^T(k) \tilde{\Phi}(k)} \le \max(1, 1/\sqrt{a_1 \lambda_{\min}(\mathcal{P})})$, which implies that

$$|\ell(k)| \leq \frac{|z_r(k)|}{\sqrt{1 + a_1 \lambda_{\min}(\mathcal{P})\tilde{\Phi}^T(k)\tilde{\Phi}(k)}} + c_5 \sum_{i=d}^{n_u+d} \|\theta(k) - \theta(k-i)\|$$
(C12)

where $c_5 \triangleq c_3 + c_4 \max(1, 1/\sqrt{a_1 \lambda_{\min}(\mathcal{P})}) > 0$.

$$\lim_{k \to \infty} \sum_{j=0}^{k} \|\theta(j) - \theta(j-i)\|^2$$

exists. Thus, Eq. (C13) implies that

$$\lim_{k\to\infty}\sum_{j=0}^k \ell^2(j)$$

exists.

Now, show that $\lim_{k\to\infty} W(\tilde{\Phi}(k)) = 0$. Since W and V are positive definite, it follows from Eq. (C8) that

$$0 \leq \lim_{k \to \infty} \sum_{j=0}^{k} W(\tilde{\Phi}(j)) \leq \lim_{k \to \infty} \sum_{j=0}^{k} -\Delta V(j) + a_1 \sigma_1 \lim_{k \to \infty} \sum_{j=0}^{k} \ell^2(j)$$
$$\leq V(\tilde{\Phi}(0)) - \lim_{k \to \infty} V(\tilde{\Phi}(k)) + a_1 \sigma_1 \lim_{k \to \infty} \sum_{j=0}^{k} \ell^2(j)$$
$$\leq V(\tilde{\Phi}(0)) + a_1 \sigma_1 \lim_{k \to \infty} \sum_{j=0}^{k} \ell^2(j)$$

where the upper and lower bound imply that all limits exist. Thus, $\lim_{k\to\infty} W(\tilde{\Phi}(k)) = 0$. Since $W(\tilde{\Phi}(k))$ is a positive-definite function of $\tilde{\Phi}(k)$, it follows that $\lim_{k\to\infty} \|\tilde{\Phi}(k)\| = 0$.

To prove that u(k) is bounded, first note that since $\lim_{k\to\infty} \|\tilde{\Phi}(k)\| = 0$ and $\Phi_*(k)$ is bounded, it follows that $\Phi(k)$ is bounded. Next, since $\Phi(k)$ is bounded, it follows from Lemma 4 that $\phi(k)$ is bounded. Furthermore, since y(k) and u(k) are components of $\phi(k+1)$, it follows that y(k) and u(k) are bounded.

To prove that $\lim_{k\to\infty} z(k) = 0$, note that it follows from Eq. (C3) and the fact that $\|\mathcal{B}z_f(k)\| = |z_f(k)|$ that

$$\lim_{k \to \infty} |z_f(k)| \le \lim_{k \to \infty} \|\tilde{\Phi}(k+1)\| + \|\mathcal{A}_*\|_F \lim_{k \to \infty} \|\tilde{\Phi}(k)\| = 0$$

Since $\lim_{k\to\infty} z_f(k) = 0$, $z_f(k) = \bar{\alpha}_m(\mathbf{q}^{-1})z(k)$, and $\alpha_m(\mathbf{q}) = \mathbf{q}^{n_m}\bar{\alpha}_m(\mathbf{q}^{-1})$ is an asymptotically stable polynomial, it follows that $\lim_{k\to\infty} z(k) = 0$.

Appendix D: Lemma used in the Proof of Theorem 3

The following lemma is used in the proof of Theorem 3. This lemma is presented for an arbitrary feedback controller given by Eq. (6) where the controller parameter vector $\theta(k)$ is time varying. More precisely, the following lemma does not depend on the adaptive law used to update $\theta(k)$ provided that such an adaptive law satisfies the assumptions in the lemma.

Lemma 4. Consider the open-loop system (1) satisfying Assumptions 1–9. In addition, consider a feedback controller given by Eq. (6) that satisfies the following assumptions:

Assumption D1. $\theta(k)$ is bounded.

Assumption D2. $\lim_{k\to\infty} \|\theta(k) - \theta(k-1)\| = 0.$

Assumption D3. There exist $\epsilon > 0$ and $k_1 > 0$ such that, for all $k \ge k_1$ and for all $i = 1, ..., n_u$, $|M(\xi_i, k)| \ge \epsilon$.

Then, for all initial conditions x_0 , there exist $k_2 > 0$, $c_1 > 0$, and $c_2 > 0$, such that, for all $k \ge k_2$, and, for all $N = 0, \ldots, n_u$, $\|\phi(k - d - N)\| \le c_1 + c_2 \|\Phi(k)\|$.

Proof. For all $k \ge n_c$, consider the $(2n_c + 1)$ th-order nonminimalstate-space realization of Eq. (6) given by

$$\phi(k+1) = \mathcal{A}_c(k)\phi(k) + \mathcal{B}_c y(k) + \mathcal{D}_r r_f(k+1)$$
(D1)

$$u(k) = \theta^T(k)\phi(k) \tag{D2}$$

where

$$\mathcal{A}_{c}(k) \triangleq \mathcal{A}_{nil} + \mathcal{B}\theta^{T}(k), \qquad \mathcal{B}_{c} \triangleq E_{2n_{c}+1}$$

Furthermore, note that, for all $k \ge n_c$, $A_c(k)$ has $n_c + 1$ poles at zero and n_c poles that coincide with the roots of $M(\mathbf{z}, k)$, which implies that, for all $k \ge n_c$, and, for all $i = 1, ..., n_u$,

$$\det(\xi_i I_{2n_c+1} - \mathcal{A}_c(k)) = \xi_i^{n_c+1} M(\xi_i, k)$$
(D3)

Next, rewrite the closed-loop system (37) and (38) as

$$\phi(k+1) = \tilde{\mathcal{A}}(k)\phi(k) + \mathcal{D}_1\psi_w(k) + \mathcal{D}_r r_f(k+1)$$
(D4)

$$y(k) = \mathcal{C}\phi(k) + \mathcal{D}_2\psi_w(k) \tag{D5}$$

where $\tilde{\mathcal{A}}(k) \triangleq \mathcal{A}_* + \mathcal{B}\bar{\theta}^T(k) = \mathcal{A} + \mathcal{B}\theta^T(k) = \mathcal{A}_{nil} + \mathcal{B}_c\mathcal{C} + \mathcal{B}\theta^T(k)$, and note that $\tilde{\mathcal{A}}(k) = \mathcal{A}_c(k) + \mathcal{B}_c\mathcal{C}$. Define the closed-loop dynamics matrix of Eqs. (D1), (D2), (D4), and (D5)

$$\tilde{A}_{\rm cl}(k) \triangleq \begin{bmatrix} \mathcal{A} & \mathcal{B}\theta^T(k) \\ \mathcal{B}_c \mathcal{C} & \mathcal{A}_c(k) \end{bmatrix}$$

Since, for all $i = 1, ..., n_u, \xi_i$ is a zero of $(\mathcal{A}, \mathcal{B}, \mathcal{C})$, it follows that $\mathcal{C}(\xi_i I_{2n_c+1} - \mathcal{A})^{-1}\mathcal{B} = 0$. Since, for all $i = 1, ..., n_u, \xi_i$ is not an eigenvalue of \mathcal{A} , it follows that $\xi_i I_{2n_c+1} - \mathcal{A}$ is nonsingular. Therefore, using Proposition 2.8.3 in [33] implies that

$$det(\xi_i I - \mathcal{A}_{cl}(k)) = det(\xi_i I - \mathcal{A}) det(\xi_i I - \mathcal{A}_c(k))$$
$$- \mathcal{B}_c \mathcal{C}(\xi_i I - \mathcal{A})^{-1} \mathcal{B} \theta^T(k))$$
$$= det(\xi_i I - \mathcal{A}) det(\xi_i I - \mathcal{A}_c(k))$$
(D6)

Next, consider the change of basis

$$\begin{bmatrix} \tilde{\mathcal{A}}(k) & \mathcal{B}\theta^{T}(k) \\ 0 & \mathcal{A}_{\text{nil}} \end{bmatrix} = \begin{bmatrix} I & 0 \\ -I & I \end{bmatrix} \tilde{\mathcal{A}}_{\text{cl}}(k) \begin{bmatrix} I & 0 \\ I & I \end{bmatrix}$$

which implies that, for all $k \ge n_c$, and, for all $i = 1, ..., n_u$,

$$det(\xi_i I - \tilde{\mathcal{A}}_{cl}(k)) = det(\xi_i I - \tilde{\mathcal{A}}(k)) det(\xi_i I - \mathcal{A}_{nil})$$
$$= \xi_i^{2n_c+1} det(\xi_i I - \tilde{\mathcal{A}}(k))$$
(D7)

Combining Eqs. (D3), (D6), and (D7) yields

$$\det(\xi_i I - \mathcal{A}(k)) = \xi_i^{-n_c} M(\xi_i, k) \det(\xi_i I - \mathcal{A})$$

and it follows from Assumption D3 that, for all $k \ge k_1$, and, for all $i = 1, ..., n_u$,

$$|\det(\xi_{i}I - \hat{\mathcal{A}}(k))| = |\xi_{i}^{-n_{c}}||M(\xi_{i}, k)||\det(\xi_{i}I - \mathcal{A})|$$

$$\geq \epsilon |\xi_{i}^{-n_{c}}||\det(\xi_{i}I - \mathcal{A})|$$
(D8)

Since $\min_{i=1,...,n_u} |\xi_i^{-n_c}| > 0$ and $\min_{i=1,...,n_u} |\det(\xi_i I_{2n_c+1} - \mathcal{A})| > 0$, it follows from Eq. (8) that, for all $k \ge k_1$, and, for all $i = 1, ..., n_u$,

$$0 < \epsilon_2 \le |\det(\xi_i I - \hat{\mathcal{A}}(k))| \tag{D9}$$

where

$$\epsilon_2 \triangleq \epsilon \left(\min_{i=1,\dots,n_u} |\xi_i^{-n_c}| \right) \left(\min_{i=1,\dots,n_u} |\det(\xi_i I_{2n_c+1} - \mathcal{A})| \right) > 0$$

Next, write $\beta_u(\mathbf{q})$ as

$$\beta_u(\mathbf{q}) = \beta_{u,0}\mathbf{q}^{n_u} + \beta_{u,1}\mathbf{q}^{n_u-1} + \dots + \beta_{u,n_u-1}\mathbf{q} + \beta_{u,n_u}$$

where $\beta_{u,0} = 1$ and $\beta_{u,1}, \dots, \beta_{u,n_u} \in \mathbb{R}$. Furthermore, for $i \ge 0$ and $j \ge i$, define

$$\begin{split} &\Lambda_{i,j}(k) \\ &\triangleq \begin{cases} \tilde{\mathcal{A}}(k-d-i-1)\tilde{\mathcal{A}}(k-d-i-2)\cdots\tilde{\mathcal{A}}(k-d-j), & j>i\\ I, & j=i \end{cases} \end{split}$$
(D10)

and define

$$\Omega(k) \triangleq \sum_{i=0}^{n_u} \beta_{u,i} \Lambda_{i,n_u}(k)$$
(D11)

It follows from Eqs. (D10) and (D11) that

$$\begin{split} \Omega(k)\phi(k-d-n_u) &= \beta_{u,0}\tilde{\mathcal{A}}(k-d-1)\cdots\tilde{\mathcal{A}}(k-d-n_u)\phi(k-d-n_u) \\ &+ \beta_{u,1}\tilde{\mathcal{A}}(k-d-2)\cdots\tilde{\mathcal{A}}(k-d-n_u)\phi(k-d-n_u) \\ &+ \beta_{u,2}\tilde{\mathcal{A}}(k-d-3)\cdots\tilde{\mathcal{A}}(k-d-n_u)\phi(k-d-n_u) \\ &+ \cdots + \beta_{u,n_u-1}\tilde{\mathcal{A}}(k-d-n_u)\phi(k-d-n_u) \\ &+ \beta_{u,n_u}\phi(k-d-n_u) \end{split}$$
(D12)

1784

Next, repeatedly substituting Eq. (D4) into Eq. (D12), and using Eqs. (10) and (D10) yields

$$\Omega(k)\phi(k-d-n_u) = \frac{1}{\beta_d}\Phi(k) - \sum_{i=0}^{n_u-1}\sum_{j=i}^{n_u-1}\beta_{u,i}\Lambda_{i,j}(k) \\ \times [\mathcal{D}_1\psi_w(k-d-1-j) + \mathcal{D}_r r_f(k-d-j)]$$
(D13)

Now, show that there exist $k_2 \ge 0$ and $\bar{\epsilon} > 0$ such that, for all $k \ge k_2$, $\Omega(k)$ is nonsingular and $\bar{\epsilon} \le |\det \Omega(k)|$. First, note that, for all $i = 0, \ldots, n_u - 1$,

$$\begin{split} \Lambda_{i,n_{u}}(k) &= \mathcal{A}(k-d-1-i)\mathcal{A}(k-d-2-i)\cdots\mathcal{A}(k-d-n_{u}) \\ &= \tilde{\mathcal{A}}(k-d-1-i)\cdots\tilde{\mathcal{A}}(k-d-n_{u}+2)\tilde{\mathcal{A}}^{2}(k-d-n_{u}) \\ &+ \tilde{\mathcal{A}}(k-d-1-i)\cdots\tilde{\mathcal{A}}(k-d-n_{u}+2) \\ &\times [\tilde{\mathcal{A}}(k-d-n_{u}+1)-\tilde{\mathcal{A}}(k-d-n_{u})]\tilde{\mathcal{A}}(k-d-n_{u}) \\ &= \Lambda_{i,n_{u}-2}(k)\tilde{\mathcal{A}}^{2}(k-d-n_{u}) \\ &+ \Lambda_{i,n_{u}-2}(k)\Delta(k-d-n_{u}+1)\Lambda_{n_{u}-1,n_{u}}(k) \end{split}$$
(D14)

where, for all $k > n_c$,

$$\Delta(k) \triangleq \tilde{\mathcal{A}}(k) - \tilde{\mathcal{A}}(k-1) = \mathcal{B}[\theta(k) - \theta(k-1)]^T$$
(D15)

Therefore, repeating the process in Eq. (D14) yields

$$\Lambda_{i,n_u}(k) = \tilde{\mathcal{A}}^{n_u - i}(k - d - n_u) + \sum_{j=1}^{n_u - i - 1} \Lambda_{i,n_u - 1 - j}(k) \Delta(k - d - n_u + j) \Lambda_{n_u - j,n_u}(k)$$
(D16)

It follows from Eqs. (D11) and (D16) that $\Omega(k) = \Omega_1(k) + \Omega_2(k)$ where

$$\Omega_1(k) \triangleq \sum_{i=0}^{n_u} \beta_{u,i} \tilde{\mathcal{A}}^{n_u - i} (k - d - n_u)$$
(D17)

$$\Omega_{2}(k) \triangleq \sum_{i=0}^{n_{u}} \beta_{u,i} \left[\sum_{j=1}^{n_{u}-i-1} \Lambda_{i,n_{u}-1-j}(k) \Delta(k-d-n_{u}+j) \Lambda_{n_{u}-j,n_{u}}(k) \right]$$
(D18)

It follows from Assumption D2 that $\lim_{k\to\infty} \Delta(k) = 0$. Since, in addition, $\theta(k)$ is bounded according to Assumption D1, it follows that for all $i \ge 0$ and for all $j \ge i$, $\Lambda_{i,j}(k)$ is bounded. Therefore, $\lim_{k\to\infty} \Omega_2(k) = 0$, which implies that there exists $\bar{k}_1 \ge 0$, such that, for all $k \ge \bar{k}_1$,

$$\frac{1}{2}|\det\Omega_1(k)| \le |\det(\Omega_1(k) + \Omega_2(k))| = |\det\Omega(k)|$$
 (D19)

Next, note that $\Omega_1(k) = (-1)^{n_u} (\xi_1 I - \tilde{\mathcal{A}}(k - d - n_u)) \cdots (\xi_{n_u} I - \tilde{\mathcal{A}}(k - d - n_u))$. Therefore, it follows from Eq. (D9) that, for all $k \ge k_1 + d + n_u$,

$$0 < \epsilon_2^{n_u} \le |\det \Omega_1(k)| \tag{D20}$$

Combining Eqs. (D19) and (D20) implies that, for all $k \ge k_2 \triangleq \max(\bar{k}_1, k_1 + d + n_u)$,

$$0 < \bar{\epsilon} \le |\det \Omega(k)| \tag{D21}$$

where $\bar{\epsilon} \stackrel{\Delta}{=} \frac{1}{2} \epsilon_2^{n_u}$, and thus $\Omega(k)$ is nonsingular.

Since, for all $k \ge k_2$, $\Omega(k)$ is nonsingular, it follows from Eq. (D13) that, for all $k \ge k_2$,

$$\begin{split} \phi(k-d-n_u) &= \frac{1}{\beta_d} \Omega^{-1}(k) \Phi(k) \\ &- \Omega^{-1}(k) \sum_{i=0}^{n_u-1} \sum_{j=i}^{n_u-1} \beta_{u,i} \Lambda_{i,j}(k) \mathcal{D}_r r_f(k-d-j) \\ &- \Omega^{-1}(k) \sum_{i=0}^{n_u-1} \sum_{j=i}^{n_u-1} \beta_{u,i} \Lambda_{i,j}(k) \mathcal{D}_1 \psi_w(k-d-1-j) \end{split}$$
(D22)

For all $N = 0, ..., n_u$, substituting Eq. (D22) into Eq. (D4) $n_u - N$ times implies

$$\begin{split} \phi(k-d-N) &= \frac{1}{\beta_d} \Lambda_{N,n_u}(k) \Omega^{-1}(k) \Phi(k) \\ &- \Lambda_{N,n_u}(k) \Omega^{-1}(k) \sum_{i=0}^{n_u-1} \sum_{j=i}^{n_u-1} \beta_{u,i} \Lambda_{i,j}(k) \mathcal{D}_r r_f(k-d-j) \\ &- \Lambda_{N,n_u}(k) \Omega^{-1}(k) \sum_{i=0}^{n_u-1} \sum_{j=i}^{n_u-1} \beta_{u,i} \Lambda_{i,j}(k) \mathcal{D}_1 \psi_w(k-d-1-j) \\ &+ \sum_{i=1}^{n_u-N} \Lambda_{N,n_u-i}(k) [\mathcal{D}_1 \psi_w(k-d-n_u+i) \\ &+ \mathcal{D}_r r_f(k+1-d-n_u+i)] \end{split}$$

which implies

$$\begin{split} \|\phi(k-d-N)\| &\leq \frac{1}{|\beta_{d}|} \frac{\|\Omega^{A}(k)\|}{|\det \Omega(k)|} \|\Lambda_{N,n_{u}}(k)\| \|\Phi(k)\| \\ &+ \frac{\|\Omega^{A}(k)\|}{|\det \Omega(k)|} \|\Lambda_{N,n_{u}}(k)\| \|\mathcal{D}_{r}\| \\ &\times \left(\sum_{i=0}^{n_{u}-1} \sum_{j=i}^{n_{u}-1} |\beta_{u,i}| \|\Lambda_{i,j}(k)\| |r_{f}(k-d-j)|\right) \\ &+ \frac{\|\Omega^{A}(k)\|}{|\det \Omega(k)|} \|\Lambda_{N,n_{u}}(k)\| \|\mathcal{D}_{1}\| \\ &\times \left(\sum_{i=0}^{n_{u}-1} \sum_{j=i}^{n_{u}-1} |\beta_{u,i}| \|\Lambda_{i,j}(k)\| \|\psi_{w}(k-d-1-j)\|\right) \\ &+ \|\mathcal{D}_{1}\| \left(\sum_{i=1}^{n_{u}-N} \|\Lambda_{N,n_{u}-i}(k)\| \|\psi_{w}(k-d-n_{u}+i)\|\right) \\ &+ \|\mathcal{D}_{r}\| \left(\sum_{i=1}^{n_{u}-N} \|\Lambda_{N,n_{u}-i}(k)\| |r_{f}(k+1-d-n_{u}+i)|\right) \end{split}$$
(D23)

where $\|\cdot\|$ is the Euclidean norm for vectors and the corresponding induced norm for matrices, and $\Omega^A(k)$ is the adjugate of $\Omega(k)$.

Since $\theta(k)$ is bounded, it follows that $\Omega(k)$ is bounded, and thus $\Omega^A(k)$ is bounded, which implies that $c_\Omega \triangleq \sup_{k\geq 0} \|\Omega^A(k)\|$ exists. Furthermore, since $\theta(k)$ is bounded, it follows that, for all $i \geq 0$, and, for all $j \geq i$, $\Lambda_{i,j}(k)$ is bounded, and thus $c_\Lambda \triangleq \sup_{i,j=0,\dots,n_u} (\sup_{k\geq 0} \|\Lambda_{i,j}(k)\|)$ exists. In addition, since $\psi_w(k)$ and r(k) are bounded, it follows that $c_\psi \triangleq \sup_{k\geq 0} \|\psi_w(k)\|$ exists and $c_r \triangleq \sup_{k\geq 0} |r_f(k)|$ exists. Therefore, it follows from Eqs. (D21) and (D23) that

$$\begin{aligned} \|\phi(k-d-N)\| &\leq \frac{c_{\Omega}c_{\Lambda}}{\bar{\epsilon}|\beta_{d}|} \|\Phi(k)\| \\ &+ (n_{u}-N)(\|\mathcal{D}_{1}\|c_{\psi}+\|\mathcal{D}_{r}\|c_{r})c_{\Lambda} \\ &+ \frac{c_{\Omega}(\|\mathcal{D}_{1}\|c_{\psi}+\|\mathcal{D}_{r}\|c_{r})c_{\Lambda}^{2}}{\bar{\epsilon}} \left(\sum_{i=0}^{n_{u}-1} (n_{u}-i)|\beta_{u,i}|\right) \\ &\leq c_{1}+c_{2}\|\Phi(k)\| \end{aligned}$$

where

$$c_1 \stackrel{\Delta}{=} (n_u - N)(\|\mathcal{D}_1\|c_{\psi} + \|\mathcal{D}_r\|c_r)c_{\Lambda} + \frac{c_{\Omega}(\|\mathcal{D}_1\|c_{\psi} + \|\mathcal{D}_r\|c_r)c_{\Lambda}^2}{\bar{\epsilon}} \left(\sum_{i=0}^{n_u-1} (n_u - i)|\beta_{u,i}|\right)$$

and

$$c_2 \triangleq \frac{c_{\Omega} c_{\Lambda}}{\bar{\epsilon} |\beta_d|}$$

References

- Monopoli, R. V., "Model Reference Adaptive Control with an Augmented Error Signal," *IEEE Transactions on Automatic Control*, Vol. 19, 1974, pp. 474–484. doi:10.1109/TAC.1974.1100670
- [2] Narendra, K. S., and Valavani, L. S., "Stable Adaptive Controller Design—Direct Control," *IEEE Transactions on Automatic Control*, Vol. 23, 1978, pp. 570–583. doi:10.1109/TAC.1978.1101823
- [3] Narendra, K. S., Lin, Y. H., and Valavani, L. S., "Stable Adaptive Controller Design, Part II: Proof of Stability," *IEEE Transactions on Automatic Control*, Vol. 25, 1980, pp. 440–448. doi:10.1109/TAC.1980.1102362
- [4] Narendra, K. S., and Annaswamy, A. M., Stable Adaptive Systems, Prentice–Hall, New York, 1989, Chap. 5.
- [5] Åström, K. J., and Wittenmark, B., Adaptive Control, Addison-Wesley, Reading, MA, 2nd ed., 1995, Chap. 5.
- [6] Ioannou, P., and Sun, J., *Robust Adaptive Control*, Prentice–Hall, New York, 1996, Chap. 6.
- [7] Tao, G., Adaptive Control Design and Analysis, Wiley, New York, 2003, Chaps. 5, 6.
- [8] Ioannou, P., and Fidan, B., Adaptive Control Tutorial, SIAM, Philadelphia, 2006, Chaps. 4, 5, 7.
- [9] Narendra, K. S., and Lin, Y. H., "Stable Discrete Adaptive control," *IEEE Transactions on Automatic Control*, Vol. 25, 1980, pp. 456–461. doi:10.1109/TAC.1980.1102365
- [10] Goodwin, G. C., and Sin, K. S., Adaptive Filtering, Prediction, and Control, Prentice–Hall, New York, 1984, Chaps. 5, 6.
- [11] Johansson, R., "Global Lyapunov Stability and Exponential Convergence of Direct Adaptive Control," *International Journal of Control*, Vol. 50, 1989, pp. 859–869. doi:10.1080/00207178908953402
- [12] Akhtar, S., and Bernstein, D. S., "Lyapunov-Stable Discrete-Time Model Reference Adaptive Control," *International Journal of Adaptive Control and Signal Processing*, Vol. 19, 2005, pp. 745–767. doi:10.1002/acs.876
- [13] Sastry, S., and Bodson, M., Adaptive Control: Stability, Convergence, and Robustness, Prentice–Hall, Englewood Cliffs, NJ, 1989, Chap. 7.
- [14] Goodwin, G. C., Ramadge, P. J., and Caines, P. E., "Discrete-Time Multivariable Adaptive Control," *IEEE Transactions on Automatic Control*, Vol. 25, 1980, pp. 449–456. doi:10.1109/TAC.1980.1102363
- [15] Hoagg, J. B., Santillo, M. A., and Bernstein, D. S., "Discrete-Time Adaptive Command Following and Disturbance Rejection with Unknown Exogenous Dynamics," *IEEE Transactions on Automatic Control*, Vol. 53, 2008, pp. 912–928. doi:10.1109/TAC.2008.920234
- [16] Bayard, D. S., "Extended Horizon Liftings for Stable Inversion of Nonminimum-Phase Systems," *IEEE Transactions on Automatic Control*, Vol. 39, 1994, pp. 1333–1338. doi:10.1109/9.293208
- [17] Annaswamy, A. M., and Wong, J.-E., "Adaptive Control in the Presence of Saturation Non-Linearity," *International Journal of Adaptive Control and Signal Processing*, Vol. 11, 1997, pp. 3–19.

doi:10.1002/(SICI)1099-1115(199702)11:1<3::AID-ACS391>3.0. CO;2-T

- [18] Fuentes, R., Hoagg, J. B., Anderton, B. J., D'Amato, A. M., and Bernstein, D. S., "Investigation of Cumulative Retrospective Cost Adaptive Control for Missile Application," *AIAA Guidance, Navigation, and Control Conference*, AIAA Paper 2010-7577, Toronto, Aug. 2010.
- [19] Yan, J., Hoagg, J. B., Hindman, R. E., and Bernstein, D. S., "Longitudinal Aircraft Dynamics and the Instantaneous Acceleration Center of Rotation," *IEEE Control Systems Magazine*, Vol. 31, No. 4, Aug. 2011, pp. 68–92. doi:10.1109/MCS.2011.941250
- [20] Santillo, M. A., and Bernstein, D. S., "Adaptive Control Based on Retrospective Cost Optimization," *Journal of Guidance, Control, and Dynamics*, Vol. 33, 2010, pp. 289–304. doi:10.2514/1.46741
- [21] Hoagg, J. B., and Bernstein, D. S., "Cumulative Retrospective Cost Adaptive Control with RLS-Based Optimization," *Proceedings* of the American Control Conference, Baltimore, MD, June 2010, pp. 4016–4021.
- [22] Hoagg, J. B., and Bernstein, D. S., "Retrospective Cost Adaptive Control for Nonminimum-Phase Discrete-Time Systems, Part 1 and Part 2," *Proceedings of the Conference on Decision and Control*, Atlanta, Dec. 2010, pp. 893–904.
- [23] Hoagg, J. B., and Bernstein, D. S., "Retrospective Cost Model Reference Adaptive Control for Nonminimum-Phase Discrete-Time Systems, Part 1 and Part 2," *Proceedings of the American Control Conference*, San Francisco, June 2011, pp. 2927–2938.
- [24] Praly, L., Hung, S. T., and Rhode, D. S., "Towards a Direct Adaptive Scheme for a Discrete-Time Control of a Minimum-Phase Continuous-Time System," *Proceedings of the Conference on Decision and Control*, Ft. Lauderdale, FL, Dec. 1985, pp. 1188–1191.
- [25] Morozov, A. V., Hoagg, J. B., and Bernstein, D. S., "A Computational Study of the Performance and Robustness Properties of Retrospective Cost Adaptive Control," *AIAA Guidance, Navigation, and Control Conference*, AIAA Paper 2010-8011, Toronto, Aug. 2010.
- [26] Hoagg, J. B., Santillo, M. A., and Bernstein, D. S., "Internal Model Control in the Shift and Delta Domains," *IEEE Transactions on Automatic Control*, Vol. 53, 2008, pp. 1066–1072. doi:10.1109/TAC.2008.921526
- [27] Campi, M. C., "The Problem of Pole-Zero Cancellation in Transfer Function Identification and Application to Adaptive Stabilization," *Automatica*, Vol. 32, No. 6, 1996, pp. 849–857. doi:10.1016/0005-1098(96)00010-6
- [28] Youla, D. C., Bongiorno, J. J., Jr., and Lu, C. N., "Single-Loop Feedback-Stabilization of Linear Multivariable Dynamical Plants," *Automatica*, Vol. 10, 1974, pp. 159–173. doi:10.1016/0005-1098(74)90021-1
- [29] Hoagg, J. B., Chandrasekar, J., and Bernstein, D. S., "On the Zeros, Initial Undershoot, and Relative Degree of Lumped-Mass Structures," *Journal of Dynamical Systems, Measurement, and Control*, Vol. 129, 2007, pp. 493–502. doi:10.1115/1.2719764
- [30] Bailey, R., and Jordan, T., "NASA Langley's AirSTAR Testbed: A Subscale Flight Test Capability for Flight Dynamics and Control System Experiments," *AIAA Guidance, Navigation, and Control Conference*, AIAA Paper 2008-6660, Honolulu, Aug. 2008.
- [31] Murch, A., "A Flight Control System Architecture for the NASA AirSTAR Flight Test Infrastructure," *AIAA Guidance, Navigation, and Control Conference*, AIAA 2008-6990, Honolulu, August 2008.
- [32] Blakelock, J. H., Automatic Control of Aircraft and Missiles, Wiley, New York, 1965, Chap. 2.
- [33] Bernstein, D. S., *Matrix Mathematics*, Princeton University Press, Princeton, NJ, 2009, Chap. 2.